

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Carballo, Jerónimo; Graziano, Alejandro; Schaur, Georg; Volpe Martincus, Christian

Working Paper Import processing and trade costs

IDB Working Paper Series, No. IDB-WP-01454

Provided in Cooperation with: Inter-American Development Bank (IDB), Washington, DC

Suggested Citation: Carballo, Jerónimo; Graziano, Alejandro; Schaur, Georg; Volpe Martincus, Christian (2023) : Import processing and trade costs, IDB Working Paper Series, No. IDB-WP-01454, Inter-American Development Bank (IDB), Washington, DC, https://doi.org/10.18235/0004752

This Version is available at: https://hdl.handle.net/10419/289955

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/3.0/igo/legalcode







IDB WORKING PAPER SERIES Nº IDB-WP-01454

Import Processing and Trade Costs

Jerónimo Carballo Alejandro Graziano Georg Schaur Christian Volpe Martincus

Inter-American Development Bank Integration and Trade Sector

February 2023



Import Processing and Trade Costs

Jerónimo Carballo Alejandro Graziano Georg Schaur Christian Volpe Martincus

Cataloging-in-Publication data provided by the Inter-American Development Bank Felipe Herrera Library

Import processing and trade costs / Jeronimo Carballo, Alejandro Graziano, Georg Schaur, Christian Volpe Martincus.

p. cm. — (IDB Working Paper Series ; 1454)

Includes bibliographic references.

1. Ports of entry-Peru. 2. Customs administration-Peru. 3. Commercial policy-Peru. 4. Tariff-Peru. 5. Peru-Commerce. I. Carballo, Jerónimo. II. Graziano, Alejandro. III. Schaur, Georg. IV. Volpe Martincus, Christian. V. Inter-American Development Bank. Integration and Trade Sector. VI. Series.

IDB-WP-1454

JEL Codes: F10, F13, F14.

Keywords: Trade Costs, Border Processing, Trade Policy.

http://www.iadb.org

Copyright © 2023 Inter-American Development Bank. This work is licensed under a Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives (CC-IGO BY-NC-ND 3.0 IGO) license (<u>http://creativecommons.org/licenses/by-nc-nd/3.0/igo/</u> <u>legalcode</u>) and may be reproduced with attribution to the IDB and for any non-commercial purpose, as provided below. No derivative work is allowed.

Any dispute related to the use of the works of the IDB that cannot be settled amicably shall be submitted to arbitration pursuant to the UNCITRAL rules. The use of the IDB's name for any purpose other than for attribution, and the use of IDB's logo shall be subject to a separate written license agreement between the IDB and the user and is not authorized as part of this CC-IGO license.

Following a peer review process, and with previous written consent by the Inter-American Development Bank (IDB), a revised version of this work may also be reproduced in any academic journal, including those indexed by the American Economic Association's EconLit, provided that the IDB is credited and that the author(s) receive no income from the publication. Therefore, the restriction to receive income from such publication shall only extend to the publication's author(s). With regard to such restriction, in case of any inconsistency between the Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives license and these statements, the latter shall prevail.

Note that link provided above includes additional terms and conditions of the license.

The opinions expressed in this publication are those of the authors and do not necessarily reflect the views of the Inter-American Development Bank, its Board of Directors, or the countries they represent.



Import Processing and Trade Costs[†]

Jeronimo Carballo University of Colorado Alejandro Graziano University of Nottingham

Georg Schaur Christian Volpe Martincus University of Tennessee and CESifo Inter-American Development Bank and CESifo

Abstract

We estimate import processing costs based on the time it takes to import. Our theory extends existing time-cost measures to account for uncertainty in import processing. We use detailed, highly disaggregated data on import processing dates and import values to provide evidence for our theory and estimate processing costs consistent with the theory. The evidence shows that our extensions to time-cost estimates are economically relevant to determine processing costs. We estimate that the tariff equivalent import processing costs is as high as 18 percent. WTO estimates suggest that the full implementation of the 2013 Trade Facilitation Agreement would reduce the time to trade by 1.5 days. In that case, processing costs would decrease to 13 percent.

Keywords: Trade Costs, Border Processing, Trade Policy

J.E.L. Classification: F10, F13, F14

[†]We thank Vanessa Alviarez, Joaquin Blaum, Bruce Blonigen, Matilde Bombardini, Julia Cajal Grossi, Davin Chor, Jonathan Eaton, Keith Head, Wolfgang Keller, Kala Krishna, Keith Maskus, Denis Novy, Roberta Piermartini, Alexandre Skiba, Jim Tybout, and Woan Foong Wong and seminar participants at TIGN/Ridge Conference, Geneva Trade and Development Workshop, ETH Zurich, Singapore Management University, National University of Singapore, IBS and at EIIT at Oregon, Midwest Meetings at IUPUI, RMET, CAF and CORE conferences. The views and interpretations in this paper are strictly those of the authors and should not be attributed to the Inter-American Development Bank, its executive directors, its member countries, or SUNAT. Other usual disclaimers also apply. Contact: jeronimo.carballo@colorado.edu, alejandro.graziano@nottingham.ac.uk, gschaur@utk.edu, and christianv@iadb.org

1. Introduction

The 2013 WTO Trade Facilitation Agreement is a major worldwide policy initiative that consists of provisions to simplify the processing of international shipments to reduce trade costs, but only limited research exists to inform such an initiative. This lack of evidence is partially due to the difficulty of measuring non-tariff barriers (Goldberg and Pavcnik, 2016). A valuable measure to evaluate the restrictiveness of non-tariff regulations employed by academic research, firms, policymakers, and international organizations is the time it takes to import.¹ For such a measure to reach its full potential to inform policy, time must be translated into cost.

To quantify processing costs requires a cost function to estimate. We use theory to derive a cost function consistent with firms' optimal management of the import process. Our starting point is that *processing times*, the time it takes to physically handle, move, and clear shipments through the port of entry is uncertain due to port congestion and conditionally random inspections. Then, firms must choose the *lead time*, the time between initiating and desired completion of a single or multiple steps in the supply chain before they know the shipments' processing performance. Short lead times save money, but run a greater risk of missed delivery obligations. Delayed shipments are costly due to late fees, reputation effects, and disruption of production processes (Boehm et al., 2019). By weighing the risk of late delivery against the cost of a slow supply chain, firms choose optimal lead times to minimize the total expected processing costs.

To estimate this expected import processing cost, we take advantage of a unique dataset consisting of highly detailed, transaction-level import data for Peru's main seaport, Callao. In particular, we observe import values across importing firms, origin countries, and products. The data also report the completion dates for various steps

¹e.g. Doing Business Trading Across Borders http://www.doingbusiness.org/data/ exploretopics/trading-across-borders/what-measured

in the import process, including vessel arrivals, unloading, storage at wharehouse, customs processing, and if customs processing involved an inspection of documents and/or the shipment.

Based on our transaction level data, we provide evidence that long unloading times lead to subsequent shorter storage times. This is consistent with our theory where firms allocate longer lead times to avoid running late in the case of a random processing delay.

Next, we estimate firms' expected import processing cost. The theory shows that we need to estimate two parameters: an import processing cost elasticity with respect to the median processing time, and a multiplier that captures costs associated with the risk of missing desired delivery dates. Both parameters need to be combined with the median processing time to evaluate the total import processing cost.

Consistent with our theory, we estimate the first parameter, the processing cost elasticity, from the import demand elasticity with respect to processing times. To do so, our theory relates import values to firms' beliefs about their median processing time. The main challenge to estimate the processing cost elasticity is that we do not observe what the firms know about the median processing time. Instead, we relate import values to observed median processing times. This potentially results in measurement error that leads to substantial bias in fixed-effect specifications (Grilliches and Hausman, 1986; McKinish, 2008). To address this bias, we generate instrumental variables based on port congestion and customs inspections. The identifying assumption is that they affect firms' anticipated median processing time and observed processing time, but do not enter the import equation otherwise. Firms may anticipate median processing times based on learning including lagged observation, and information they obtain from shipping and logistics companies. We examine these specific mechanisms with robustness checks.

The second parameter, the cost multiplier, is not recoverable from fixed-effect regressions that relate import values to processing times. We use our detailed import processing information to determine this parameter and provide several robustness checks. Instrumental variable estimates show that a 1 percent increase in the median processing time lowers import values by .24 percent.² Given an import demand elasticity of 4 (Soderbery, 2015), our theory translates the import processing time elasticity of .24 into a processing cost elasticity of .06. Hence, a 1 percent increase in the median time raises import processing costs by .06 percent. For the cost multiplier, we obtain an estimate of 1.104 and we provide evidence that it significantly affects import processing costs based on bootstrapped standard errors.

Evaluated at a median processing time of three days, our estimates combine to result in an import processing cost that equals a 18 percent import tariff.³ Hopes for the 2013 Trade Facilitation Agreement are high.⁴ Based on our estimates this is justified. WTO estimates suggest that the full implementation of the Trade Facilitation Agreement may result in a reduction of the time to import of 1.5 days.⁵ Our estimates imply that this would reduce import processing costs from about 18 to 13 percent.⁶ For comparison, average import tariffs worldwide equal about 5 percent in 2017.⁷

Existing literature shows that long delivery times reduce trade and increase costs (Persson, 2008; Djankov et al., 2010; Freund and Rocha, 2011; Hummels and Schaur, 2013; Volpe Martineus et al., 2015; Heid et al., 2017; Oberhofer et al., 2018; Fernandes et al. 2021). This literature assumes empirically convenient functional forms and relates import and export values to various measures of the time it takes to trade.⁸ We combine

²This elasticity is somewhat lower compared estimates in the existing literature based on export processing. For example, Djankov et al. (2010) estimate that a 1 percent increase in the time it takes to deliver a shipment from the factory gate to the port lowers trade by 0.4 percent. Instrumenting raises the magnitude of our elasticity estimates by a factor of five compared to OLS. This increase is comparable to existing IV applications (Costinot et al., 2012; Paravisini et al., 2015).

³Consistent with our theory it is simply computed according to the log linear cost function $\lambda \times (MedianProcessingTime)^{\chi} - 1 = 1.104 \times 3^{.061} - 1 = 0.18$ where λ is the cost multiplier and χ is the processing-time cost elasticity.

⁴Roberto Azevedo, former Director General of the WTO, noted that "The impact will be bigger than the elimination of all existing tariffs around the world." https://www.wto.org/english/news_e/ news17_e/fac_31jan17_e.htm

⁵https://www.wto.org/english/news_e/news17_e/fac_31jan17_e.htm

⁶Formally, $\lambda \times (MedianProcessingTime)^{\chi} - 1 = 1.104 \times 1.5^{.061} - 1 = 0.132.$

⁷https://data.worldbank.org/indicator/TM.TAX.MRCH.WM.AR.ZS?

 $^{^{8}}$ We follow this literature and develop time costs as a ad-valorem tariff equivalent. For a discussion

theory and data to extend and improve our understanding of these elasticity and timecost estimates. Our theory shows that available elasticities likely do not apply to evaluate import processing costs because they depend on the shape of the processing distribution. Our empirical results provide evidence that our extensions of existing cost estimates to account for uncertainty are economically meaningful and relevant to evaluate policy.

Our uniquely detailed import processing data has several advantages to accomplish this. These data allow us to compute a measure of processing time consistent with the theory, estimate the time cost multiplier that cannot be identified from import regressions, and, generally, provide evidence for the theory. In addition, we leverage these data for multiple robustness checks.

Who gains from trade facilitation is policy relevant. Often the hope is that small firms and new relationships will benefit and grow.⁹ However, without cost estimates, it is a priori not clear how high processing costs are for these firms and relationships, and it is difficult to measure policies' ability to reduce these costs. We provide evidence that experienced importers incur a processing tariff of about 12 percent. New importers pay a processing cost tariff equivalent more than double compared to experienced firms. This is evidence that border-related processing costs are especially relevant to the formation of new trade relationships (Bernard et al., 2017a, 2017b; Fitzgerald et al., 2017; Rodrigue and Tan, 2019).

The next section provides background information on import processing and import processing times. Section 3 develops a theory for expected costs of import processing. Section 4 introduces our detailed import data. Section 5 explains how we identify the effect of processing times on imports and reports estimates. Section 6 develops estimates for border processing costs based on the estimation results in Section 5. Section 7 presents the results of several robustness checks. Section 8 examines the heterogeneity of border

of identification of per-unit costs versus ad-valorem costs, see Irarrazabal et al. (2015). ⁹https://www.wto.org/english/news_e/news17_e/fac_31jan17_e.htm

processing costs according to existing theory, and Section 9 concludes.

2. Import Processing at the Border

In this section, we describe the import process at Peru's main seaport, Callao.¹⁰ We use highly disaggregated data taken from customs import declarations and load manifests over the period 2007-2013 kindly provided by Peru's National Tax Agency (Superintendencia Nacional de Administración Tributaria - SUNAT).

For each shipment, we observe the date when the ship arrived (arrival date), the date the shipment was unloaded and moved to the shipyard or warehouse (unload date), the date the customs import declaration was created and registered (declaration date), and the date the shipment was released by customs (release date). For the majority of shipments, over 90 percent, these dates follow a simple order:

arrival date
$$\Rightarrow$$
 unload date \Rightarrow declaration date \Rightarrow release date (1)

The remaining 10 percent of the shipments enter an express channel. In the express channel, firms may submit a customs declaration before the vessel arrives in port to expedite customs processing. Therefore, in the express channel the customs declaration date occurs first and the order of the dates changes to:

declaration date
$$\Rightarrow$$
 arrival date \Rightarrow unload date \Rightarrow release date (2)

Based on the dates we observe in our data, and taking into account the express channel, we measure the time it takes to clear three consecutive steps to import a shipment: unloading, storage, and customs processing.¹¹

¹⁰ The seaport of Callao represents over 70% of Peru's import value in a given year. See Table 3 for more detailed information.

¹¹Unfortunately, our customs data sets does not include information after the release of the shipments.

Unloading time is the number of days between the arrival date of the vessel and the date when the shipment was unloaded and moved to the shipyard or warehouse: unloading time = unloading date - arrival date. The time it takes to complete this step depends on port capacity, shipping companies, equipment failure, port strikes, etc.

Storage time is the number of days the shipment idles in port between unloading and customs processing. For shipments that do not use the express channel, see (1), the storage time is then the time between the unloading date and the declaration date: storage time = declaration date – unloading date. Therefore, the longer the firm waits to submit the customs declaration, the longer is the storage time. For shipments that use the express channel, see (2), the customs declaration is submitted before the vessel arrives at the port, and we assume that customs processing begins immediately after unloading. The reasoning is that customs inspections cannot begin until the shipment is unloaded, and firms that use the express channel do not let shipments idle in storage. Therefore, for the express channel, we set the storage time to zero: storage time = $0.^{12}$ For all shipments, storage time depends on firms' own preference on when to submit the customs declaration.

Customs processing time is the number of days shipments spend with customs. For shipments that are not express, the customs processing time is the number of days between the date the customs declaration was submitted and the date the shipment was released: customs processing time = release date – declaration date. For express shipments, customs processing begins immediately after the date of unloading of the shipment: customs processing time = release date – unload date.

For express and regular shipments, a risk management model randomly allocates shipments to one of the three customs verification channels that determine the intensity of the customs inspection and the associated duration of the customs process.¹³ The green

¹²We will perform multiple robustness checks dropping express shipments. The results do not change. ¹³Shipments are allocated according to administrative, fiscal, and security risk factors that include

channel clear shipments without inspection. The orange channel inspects documents. The red channel performs an often time consuming physical inspection.¹⁴

Based on unloading, storage, and customs processing times, we define two measures of border time. First, *total border time* is simply the sum of the unloading, storage, and customs processing times.

total border time = unloading time + storage time + customs processing time

Total border time potentially depends on firms' actions because firms decide when to initiate the customs process and the duration of the storage step.

Second, *processing time* is the sum of the unloading and customs processing time measures the time each shipment takes in the necessary processing steps.

processing time = unloading time + customs processing time

Contrary to total border time, the customs processing time depends only on actions of shipping companies, port, and border agencies and is out of the hands of individual firms.

Table 1 presents percentiles of our time measures by customs verification channel for all shipments clearing through the seaport of Callao in 2013.¹⁵ The percentiles show that total border times are a combination of official and necessary processing times of shipments, as well as a substantial amount of storage time. About 50 percent of all shipments are processed in 3 days or less (for comparison, the median total border time including storage is 10 days), but Figure 1 shows that there is a long right tail of the

the exporting firm, origin country, transport mode, transport company, countries of intermediate stops, customs broker, customs branch, product, and importing firm in Peru.

¹⁴No more than 15% of the DUAs numbered in a given month in Callao can be subject to material control (see SUNAT, 2010). SUNAT charges small fees for moving, opening, unloading, and reloading of containers. These services cost an average of 40 US dollars each.

¹⁵Data in other years are very similar. Detailed tables are available upon request. We count 1 day for stages cleared within the same day.

processing-time distribution. The percentiles in Table 1 show that processing times are associated with the (conditionally) random assignment of each shipment to a customs channel, as the processing time systematically increases with the scrutiny of the customs inspection. Dropping shipments that use the express channel does not affect the averages or the percentiles (see Table A2).¹⁶

If firms absorb random shocks in the physical handling of shipments with shorter storage times to meet contractual delivery dates, then we expect that longer processing times result in shorter storage times. To provide evidence, we take advantage of the timing of the import steps and regress log storage times on log unloading times at the transaction level.¹⁷ This relationship avoids potential issues of reverse causality, because the entire storage time we observe is after the unloading step.¹⁸ In addition, various combinations of firm, product-origin, and day of the week fixed effects account for potentially omitted variables. Conditional on this identification strategy, results may be interpreted as causal. The top panel of Table 2 shows that longer unloading times are absorbed by shorter storage times. Table A3 reports the result where we drop express shipments from the sample. The conclusions remain the same. Conditional on fixed effects absorbing omitted variables and the avoidance of reverse causality, these results provide evidence that firms use storage time to buffer against random unloading shocks.

The theory we develop in the next section formalizes this idea. It takes into account that the processing time, i.e., unloading plus customs clearance, is random and that firms allocate longer lead times to buffer processing shocks.

¹⁶The working paper Carballo et al. (2016), provides additional statistics that show that shipments of small firms and new importers are associated with longer border times. https://publications.iadb. org/publications/english/document/Endogenous-Border-Times.pdf.

¹⁷We focus on data from 2013 for the ease of exposition. We also run these regressions on the other years of our data and the results are the same.

¹⁸Alternatively, we may relate storage time to total processing time or its other component, i.e., customs processing times. However, the latter are realized after the storage stage. We did examine if longer storage times predict shorter customs processing times because better preparation may lead to faster processing. We did not find evidence that supports this mechanism.

3. Theory

3.1. Expected Total Border Entry Costs

International trade involves both physically moving shipments and administrative steps regulated by governments such as customs procedures. To allow for these procedures, firms choose a shipment's lead time by placing shipments in advance of their desired delivery dates. Short lead times save money, but they increase the expected costs associated with missing the delivery date. We model this trade-off focusing on border procedures consistent with the empirical results in Section 2, but our approach extends to the entire supply chain.

Let v denote the total import value including transportation costs, tariffs, and insurance. Let $t_l > 1$ be the lead time that firms allow for shipments to clear import procedures. Slow supply chains are costly. Therefore, let the cost of a greater lead time to clear import procedures, $t_l^{\vartheta}v$, be proportional to the shipment value, v, and increasing with constant elasticity, $\vartheta > 0$.¹⁹ If actual processing times are deterministic, then firms choose a lead time equal to the processing time and $\vartheta > 0$ captures a log-linear time cost elasticity similar to what is currently estimated in the existing literature (e.g. Djankov et al., 2010; Volpe Martincus et al., 2015).

However, processing times are random due to equipment failure, congestion, and customs inspection, and, according to Figure 1, distributed with a long right tail.²⁰ Following this pattern, let the actual processing time t_p be Pareto distributed $t_p \sim \frac{\varphi t_{min}^{\varphi}}{t_p^{\varphi+1}}$ with support $[t_{min}, \infty)$ and shape parameter $\varphi > 1$ to ensure a finite mean.²¹ For a given location of the processing distribution determined by t_{min} , a greater shape parameter φ increases the probability that the processing time t_p is less than some pre-determined lead time t_l .

¹⁹Greater lead times in our model are similar to greater time costs of money.

²⁰It is difficult to examine the distribution for each importer-exporter-product combination. In those cases with sufficient observations the distributions show a long tail similar to the overall distribution.

²¹We also solved our model with a general import processing-time distribution. However, in that case, we do not obtain a parametric cost function to estimate.

If the processing time turns out shorter than planned, $t_p \leq t_l$, then the firm stores the shipment until the desired delivery date at zero additional cost.²² This is in line with the evidence in Section 2, shorter unloading times result in longer storage times. Late shipments, $t_p > t_l$, accrue container demurrage, late fee penalties, and supply chain management costs. These costs, $(t_p/t_l)^{\omega} rv$, increase in the proportion by which the processing time exceeds the lead time as a factor of the import value, v > 0. The parameters ω and r determine the level and elasticity of these costs. Taking these costs into account, firms consider the total expected cost of importing:

$$ETC(t_l) = \int_{t_l}^{\infty} \left(\frac{t_p}{t_l}\right)^{\omega} \left(\frac{t_{min}}{t_p}\right)^{\varphi} \frac{\varphi r v}{t_p} dt_p + t_l^{\vartheta} v.$$
(3)

Allowing for more lead time, t_l , lowers the probability of missing the delivery date and expected costs of late arrivals, but raises time costs $t_l^{\vartheta} v$ due to slower supply chains.²³ Firms choose the optimal lead time, t_l^* , to minimize expected total costs of import processing:

$$t_l^* = \min_{t_l} ETC(t_l) = t_{min}^{\frac{\varphi}{\varphi+\vartheta}} \left(\frac{r}{(\varphi-\omega)}\frac{\varphi^2}{\vartheta}\right)^{\frac{1}{\varphi+\vartheta}}$$
(4)

Equation (4) shows that lead time is determined by distribution parameters (t_{min}, φ) and cost parameters (ω, r, ϑ) and we prove our proposition.²⁴

Proposition 1. For interior solutions and a given processing-time distribution, firms sched-

²²From a theory point of view, whether shipments are stored or delivered early is irrelevant as long as the costs associated with lead times are not refundable. However, we observe storage time in our data. Furthermore, from conversations with logistics companies serving the port of Callao we understand that storage is free up to 19 days. Therefore, this modeling assumption is reasonable for us. Extending the model to include additional storage costs is feasible in case this is relevant to consider different ports.

²³The processing shock t_p may be the sum over multiple steps in the supply chain. In that case, it is a caveat that we do not derive predictions on where in the supply chain firms allocate storage. A reasonable interpretation is that firms allocate storage at the end of the import process.

²⁴For interior solutions, we require $r > \vartheta(\varphi - \omega)t_{min}^{\varphi}\varphi^{-2}$ or that firms care enough about late delivery costs such that they choose $t^* > t_{min}$. In the expected cost function, the cost elasticity and shape parameter combine to the restriction that $\omega - \varphi - 1 < -1$ for the integral on the expected time cost to exist, as standard in the Pareto distribution. This results in the parameter restriction $\varphi > \omega$.

ule longer lead times t_l^* if (i) late fees are more elastic in missing the delivery date (i.e. if ω increases), (ii) late fees are a greater proportion of the import value (i.e. if r increases), (iii) if lead time costs are less elastic (i.e. if ϑ decreases).

For proof see Section A.1.

Proposition 1 has implications for the cross-country evaluation of trade facilitation measures based on processing times. Two countries' processing time distributions may be identical, but lead times and expected costs associated with import processing differ. Therefore, simple comparisons of processing time distributions, in our case t_{min} and φ , do not necessarily result in cost rankings of import processing. Thus, for data on the time it takes to import to be fully informative for policy, we must translate it into cost.

Proposition 1 also shows that comparing border processing performance based on total border times can be misleading. For two ports of entry with the same processing-time distribution, firms allow longer lead times if lead time costs are less elastic perhaps due to differences in available storage space. Therefore, longer lead times may not be a sign that processing costs are high, but that storage space is cheap.²⁵ In this case, longer lead times are associated with an increase in trade. This result emphasizes the importance of measuring effects of trade facilitation and import processing costs based on the fundamentals of the processing distribution.

We make two steps to translate import-processing times to costs. First, substitute t_l^* into (3) to obtain minimized expected costs as a function of the minimum processing time. Second, based on the Pareto distribution, substitute $t_{min} = T/\sqrt[q]{2}$, to obtain total

²⁵Applying the envelope theorem to equation (3) in optimum, $ETC(t_l^*)$, it is straightforward to see that $\frac{\partial ETC(t_l^*)}{\partial \vartheta} > 0$ as long as the processing times take at least one day, $t_p > 1$, and we are at an interior solution.

minimized expected costs as a function of the median processing time T:

$$ETC = \lambda T^{\chi} v, \tag{5}$$

where $\chi = \varphi \vartheta / (\varphi + \vartheta)$ and

$$\lambda\left(\frac{r}{\varphi-\omega},\varphi,\vartheta\right) = \left(\frac{r}{\varphi-\omega}\right)^{\frac{\vartheta}{\vartheta+\varphi}} \left(\vartheta^{\frac{\varphi}{\vartheta+\varphi}}\varphi^{-\frac{\varphi-\vartheta}{\vartheta+\varphi}} + \vartheta^{-\frac{\vartheta}{\vartheta+\varphi}}\varphi^{\frac{2\vartheta}{\vartheta+\varphi}}\right)2^{-\frac{\vartheta}{\varphi+\vartheta}} \tag{6}$$

The multiplier, λ , median processing time, T, and elasticity, χ , combine to define the border-processing cost factor, λT^{χ} , as an ad-valorem tariff equivalent on the total import value, v.

Trade facilitation policy emphasizes costs associated with slow shipment processing due to regulations of international commerce. Equation (5) then highlights potential benefits of trade facilitation policy. The elasticity χ and the multiplier λ translate policy driven reductions in median processing times, T, into lower border processing costs. To understand what determines the benefits of trade facilitation policy, we may further examine the fundamentals of χ and λ .

According to equation (5), the processing-time cost elasticity, $\chi = \varphi \vartheta / (\varphi + \vartheta)$, increases in ϑ and φ . Therefore, processing costs are more elastic with respect to a percentage change in median processing times, if lead time costs are more elastic (a greater ϑ) and the processing time distribution is subject to less probability of long delays (a greater φ) due to a steeper processing-time distribution. In addition to providing fundamentals for existing elasticity estimates, this has an important consequence for the evaluation of import-processing costs. The processing distribution, including φ , is determined by local regulations, port procedures, equipment failures, and risk management methods. Therefore, to evaluate import-processing costs, we cannot rely on existing elasticity estimates based on data from different countries, modes of transport, and legs of the international supply chain (Djankov et al., 2010; Hummels and Schaur 2013; Volpe Martincus et al., 2015; Fernandes et al., 2019). Instead, we must estimate our own processing cost elasticities. To do so, we follow the existing literature and relate processing times to trade flows.

In addition to the shape parameter, φ , and the lead time cost elasticity, ϑ , the multiplier λ also depends on costs of supply chain disruptions and late fees collected in r and ω . For a given shape of the processing distribution and lead time cost elasticity a greater cost of supply chain disruptions, an increase in r or ω , raises the multiplier λ and the expected border processing costs.²⁶ Consequently, the multiplier λ captures costs associated with missing desired delivery windows that are not included in the processing cost elasticity χ .

Estimating border processing costs requires an empirical strategy for λ . Taking advantage of the structure of our model and detailed data, we provide a estimation strategy for $r/(\varphi - \omega)$ to obtain estimates for λ . In particular, conditional on the processing-time distribution and elasticity parameters (φ, ϑ) , equation (4) shows that firms choose a greater optimal lead time t_l^* the greater $r/(\varphi - \omega)$. Therefore, information on the processingtime distribution, elasticities, and a proxy for the optimal lead time determine a value for $r/(\varphi - \omega)$ from our data. To determine the remaining parameters in λ , (φ, ϑ) , we examine the elasticity χ .

3.2. Import-Processing Cost and Imports

To link import-processing times to import values, let us focus on a given importerexporter relationship.²⁷ Firm *i* imports m_{ihxy} units of product *h* from country *x* in year

²⁶If $\lambda > 1$, then import processing is costly even if the median processing time is one, T = 1. The intuition is that even in that case where firms at the median do not experience delays, they take into account the probability of experiencing a delay and the associated costs of missing the delivery window determined by the parameters r, ω and the probability distribution, as well as the costs of hedging against such delays by scheduling longer lead times determined by ϑ .

 $^{^{27}}$ Bernard et al. (2017a) model the endogenous sorting of importers and exporters. This is beyond our object in this paper and we take an importer-exporter relationship as given. Nevertheless, we derive a log-linear import value relationship similar to their theory.

y. The firm combines the imported product with a domestic input, l_{ihxy} .²⁸ Output, q_{ihxy} , is produced and distributed according to the Cobb-Douglas production function $q_{ihxy} = \alpha_{ihx} \times \alpha_{iy} \times m_{ihxy}^{\beta} \times l_{ihxy}^{1-\beta}$. We maintain $0 < \beta < 1$. The productivity parameters α_{ihx} and α_{iy} allow for heterogeneity in productivity across importers, origin, products, and time.²⁹ Final products are differentiated and demand on the domestic market follows CES, $q_{ihxy} = A_y \left(p_{ihxy}^f \right)^{-\sigma}$. Final goods producers are monopolistically competitive on output markets and optimally source the local and international input taking prices as given.

Domestic factor markets are competitive such that the price of the domestic input, w_{hy} , varies across products and time, but not across firms. Let p_{hxy} be the f.o.b. price of the imported input and $\tau_{hxy} > 1$ be the ad-valorem import cost factor including freight and tariffs.³⁰ Taking into account import processing costs, an importer's profit maximizing³¹ import demand then is $m_{ihxy} = \kappa_{iy} \times \kappa_{ihx} \times \kappa_{hy} \times (\lambda T_{ihxy}^{\chi})^{-\gamma} \times p_{hxy}^{-\gamma}$, where the constants κ_{ihx} , κ_{iy} , κ_{hy} absorb productivity and demand parameters and $\gamma = \beta(\sigma - 1) + 1$. The exporter produces a differentiated variety with constant marginal cost z_{hxy} , takes the importers demand as given and charges the profit maximizing constant markup over marginal cost price $p_{hxy} = \frac{\gamma}{\gamma - 1} z_{hxy}$. Combining import demand with the exporter's pricing rule the import value equals

$$v_{ihxy} = m_{ihxy} p_{hxy} = \kappa_{iy} \kappa_{ihx} \delta_{hxy} \times \left(\lambda T^{\chi}_{ihxy}\right)^{-\gamma}.$$
(7)

³⁰In the empirical section we discuss how our identification strategy extends to the case where export prices vary across importers p_{ihxy} , and we provide robustness checks considering exporting firms.

³¹Firms maximize expected profits:
$$A_y^{\frac{1}{\sigma}} \left(\alpha_{ihx} m_{ihxy}^{\beta} l_{ihxy}^{1-\beta} \right)^{1-\overline{\sigma}} - \lambda T_{ihxy}^{\chi} \tau_{ihxy} p_{ihxy} m_{ihxy} - w_{hy} l_{ihxy}.$$

²⁸Note that the local factor l_{ihxy} has a x subscript. This is to distinguish that a firm may import the same product from multiple source countries and allocates some labor to finish and distribute each of these products on the market.

²⁹Alternative sourcing modeling assumptions, such as CES production, result in similar log-linear import demand functions. For example, see Halpern et al. (2015), Gopinath and Neiman (2015) and Antràs et al. (2017). In that case we can think of firms importing varieties to combine to a single output according to a CES production function, but we would obtain a similarly log-linear import equation.

The constant δ_{hxy} now accounts for demand in the importing country as well as the exporter's marginal cost. The processing costs parameters λ and χ translate an increase in the median processing time into an increase in processing cost. The parameter γ translates this cost increase into a reduction in trade flows.

In the following sections, we take advantage of equation (7) to estimate import elasticities, $\gamma \chi$, and to back out estimates for χ . The time cost multiplier, λ , is not separable from other constants in this log-linear demand equation. We use our elasticity estimates to develop an alternative estimation strategy to determine λ . Before we explain how we obtain elasticity estimates and determine import processing costs, the next section explains the import data we use throughout the rest of the paper.

4. Data

To implement equation (7) empirically requires data on imports and border processing. In this section, we discuss data sources and summary statistics.

We observe highly detailed import data obtained from Peru's National Tax Agency, SUNAT, from 2007 to 2013. Our dataset reports import values, quantities in kilograms, freight, and tariff charges for each recorded transaction. In addition, for each record we see the ID of each importing firm, the origin country of the flow, the exporting firm, the product code (10-digit HS), the customs office clearing the shipment, and the vessel that carried the shipment. These data cover all transactions entering Peru. We merge these import data with our detailed information on processing times we observe for the port of Callao described in Section 2 at the transaction level and generate an estimation sample to identify the import demand equation.³²

Before doing so, Table 3 compares the universe of import transactions for Peru with the sample of imports that arrive at the seaport of Callao. Imports clearing Callao

 $^{^{32}}$ We do not lose data due to this merge since we have transaction IDs that connect processing data with customs data.

account for approximately three quarters of the total import value, two thirds of the total number of importers, and 90% or more of all imported products and countries of origin. We therefore capture most of Peru's imports. An advantage of focusing on Callao is that the majority of business activity is concentrated around Lima which mitigates concerns that heterogeneity in inland transportation impacts our results. Furthermore, the Callao-average importer is similar to the national-average importer. More specifically, the Callao-average importer has 65 employees, is eight years old, and buys 12.4 products from 2.8 countries for approximately 650,000 US dollars (See Table A1 in the appendix for details).³³

There are 22 customs offices in Peru, but the average firm uses only 1.03 customs offices and does not appear to use multiple ports of entry in response to port congestion, long queues at customs, or other delays. Consequently, imports arriving at Callao represent the majority of the firm's imports. Therefore, merging the processing information at Callao with the firm's import information is akin to merging the firm's total imports with its processing data.

We aggregate firms' import data processed at the seaport of Callao to the importerproduct-origin-year level. Similarly, using the shipment level processing data for the port of Callao described in Section 2, and applying our definition of processing times, we generate median processing times, \hat{T}_{ihxy} , across all shipments within each importerproduct-origin-year unit of observation. In addition to median processing time, we will also use a measure of the total border time. Section 2 defines the total border time for each shipment as the difference in days between the date when the vessel arrives and the shipment clears from customs. For the following empirical sections, we employ the median

³³The national-average importer has 52 employees, is seven years old, and buys 14 products from 3.1 countries for roughly one million US dollars (See Table A1 in the appendix for details). Hence the Callao-average importer looks like the national-average importer, but imports less in terms of value spread over a smaller number of shipments. The difference are due to heavy goods being imported through other ports located closer to the production facilities and imports entering through airports which typically consists of smaller and more frequent transactions (see Table A1 in the appendix).

of the total border time across all shipments within each importer-product-origin-year unit of observation as our measure of total border time.

Aggregation to the importer-product-origin-year level facilitates standard empirical approaches. For example, it is straight forward to account for time varying fixed effects and use lagged variables to achieve identification. This is much more challenging in transaction level data where shipments across different importers, exports, and products arrive on different days resulting in much noisier variation. While convenient, this aggregation results in a seeming disconnect between our theory and data. Our theory is based on an importer-exporter relationship, but for most of our empirical applications, we aggregate to the importer-product-origin-year level. Aggregating across exporting firms is relatively inconsequential. Within an importer-product-origin-year combination, Peruvian firms tend to source only from a few exporters. Nevertheless, we will examine this with a robustness check. Aggregating to annual observations sums over multiple shipments within the year. Our theory is mute on the frequency of shipments, but we also examine potential consequences of this aggregation with robustness checks.

Combined, we have an estimation sample that includes f.o.b. import values (v_{ihxy}^{fob}) , freight charges, tariffs, insurance charges, the median processing time, \hat{T}_{ihxy} , and a measure of the median total border time within each importer-product-origin-year unit of observation. See Table A4 for descriptive statistics on these variables computed using our main estimation sample.

5. Identification of Import Elasticities w.r.t. Border Processing Time

With detailed import data at hand, in this section we explain how we identify the effect of import-processing times on imports. We develop the empirical model, discuss the identification strategy, and report baseline results. We discuss robustness checks in a later section.

5.1. Empirical Specification and Identification

We take equation (7) to our data to estimate $\gamma \chi$. This presents a challenge. In our theory, firms know the median processing time, T_{ihxy} . Unfortunately, we do not see what firms know, but observe realized median processing times, \hat{T}_{ihxy} , for importer *i* across products *h*, origin of exports *x*, and within each year *y*. To bridge this gap, we apply a proxy variable approach. Let actual processing performance equal a firm's beliefs regarding shipment processing time plus a random shock such that, $ln\hat{T}_{ihxy} =$ $lnT_{ihxy} + e_{ihxy}$ where $E(e_{ihxy}) = 0$. Then, taking logs of the import value equation, equation (7), and substituting the proxy \hat{T}_{ihxy} for the unobserved information T_{ihxy} we obtain the empirical model:

$$ln(v_{ihxy}) = \delta_{hxy} + \kappa_{iy} + \kappa_{ihx} + \gamma \chi ln T_{ihxy} + u_{ihxy}, \tag{8}$$

where the disturbance u_{ihxy} contains measurement error e_{ihxy} . The main parameter of interest is $\gamma \chi < 0$. The empirical model shows that log-linear specifications, the common approach in this literature, implicitly fix the shape of the processing-time distribution, φ , within the elasticity χ . We follow the literature and treat $\gamma \chi$ as a parameter to estimate.³⁴ Importer-year fixed effects account for firm-level changes in productivity. Importer-product-origin fixed effects, κ_{ihx} , absorb heterogeneity in importer-exporter relationships. Product-origin-year fixed effects, δ_{hxy} , account for exporter productivity and changes in supply as well as trade policy conditions.³⁵

Our identification strategy depends on the assumption that firms have knowledge about the median processing time T_{ihxy} and that this median processing time is related

³⁴The alternative is to treat the shape parameter as data. This would require a non-linear identification strategy that accommodates a large number of fixed effects, avoids the incidental parameter problem, and handles instrumental variables to break endogeneity. We are not aware of a convenient estimator to handle these challenges.

³⁵These fixed effects also account for the seasonality of products. Furthermore, we assume that firms take the shipment schedule as given. If firms have means to reduce the cost of delays by adjusting their shipment schedule, then we expect that this reduces the elasticity of imports with respect to delays.

to the observed processing time \hat{T}_{ihxy} . There are several sources of information that firms may use to gain knowledge about the median processing time. They may obtain information from carriers and logistic companies, they may use past experience, and there may be learning from individual shipments within annual observations, i.e., firms that import frequently are able to observe year-to-year changes in processing times within the calendar year, and are thus able to update their priors about current processing times and adjust their import decisions accordingly. We will take advantage and examine all of these mechanisms. If firms do not have information about the median processing time which they can use to evaluate import processing costs, or, if their priors about the median processing costs are false and are not related to observed processing times, then we expect $\hat{\gamma}\chi = 0$.

The main challenge to obtain a consistent estimate for $\gamma \chi$ is that measurement error, e_{ihxy} , is contained in the disturbance and is correlated with the main regressor of interest, $ln\hat{T}_{ihxy}$, according to our proxy variable approach.³⁶ In fixed-effects regressions, classical measurement error is known to lead to substantial attenuation bias because variation of the independent variable around the fixed effects usually emphasizes variation in idiosyncratic measurement error (Griliches and Hausman, 1986; Mckinish, 2008). We develop two instruments based on port congestion and inspection probabilities to solve this problem. We will first introduce the instruments and then discuss their necessary identification assumptions.

Simultaneous arrival of several vessels translates into longer border handling and processing times due to congestion. In our data, we observe the arrival date of each vessel and use it to compute the number of vessels that arrived the day before each shipment. Then, for each importer-product-origin-year combination, we take the median of this measure

³⁶The alternative is to make the much more convenient assumption that $T_{ihxy} = \hat{T}_{ihxy} + e_{ihxy}$ and e_{ihxy} is not systematically related to \hat{T}_{ihxy} . In that case, OLS is consistent and we would expect that IV and OLS estimates are similar, unless there are additional sources of bias.

across all shipments as a measure of congestion and our first instrument. Consequently, even if we aggregate to annual frequencies, this measure exploits within year variation of arrival dates of shipments. For example, for two shipments of the same product from the same origin the congestion measure may differ if they are imported on different dates within the year. As a consequence, for a given time period, the congestion instrument varies across importers, products and origins. ³⁷

Our second instrument is based on the fact that handling time in customs depends on the assignment to different inspection channels. A customs' risk management model allocates shipments to different processing channels. Some shipments pass customs without further inspection. Other shipments experience additional processing burden due to document and physical inspections. Within each importer-product-origin-year observation, we compute the fraction of shipments that were assigned to more intensive inspection channels. We focus on this unit of observations, because the customs risk management model takes into account, firm, product, and origin information. As a result, the instrument captures the exogenous probability of assignment to more time-consuming inspection channels across time and importer-product-origin triplets within the same period. We examine sensitivity of our results with respect to alternative definitions of the instruments, including lagging the instruments, in the robustness section.

The instruments must predict realized median processing times, $ln\hat{T}_{ihxy}$. This is easily verifiable from first stage statistics. The instruments must also not be related to the outcome in specification (8) after conditioning on the other explanatory variables. To achieve this, we absorb omitted variables that may be correlated with inspection probabilities and port congestion with fixed effects. If customs selects inspection probabilities based on relationship specific information, or, based on the origin country and product,

³⁷A concern might be that importers would bring large shipments right before the Christmas shopping season, which is exactly the time of high congestion in the port. This is challenging to examine at high frequency, because of lumpy shipments in international trade. However, we estimated our baseline model at quarterly frequency and report the results as a robustness check in the appendix. In general, they confirm our findings.

then firm-product-origin fixed effects, κ_{ihx} , and product-origin-year fixed effects, δ_{hxy} , account for this information. These fixed effects also account for changes in product specific demand, heterogeneity in the sophistication of existing supply chains such as the use of information technology, and distance related transportation costs. Finally, firm-year fixed effects, κ_{iy} , control for firms' size, experience, importer-year specific productivity, and firms' supply chain complexity. In the robustness section, we will also account for heterogeneity across carriers and exporting firms.³⁸ This mitigates concerns that high performing imports are associated with better logistics providers that result in reduced import processing.

Finally, according to the theory, specification (8) accounts for export prices and freight charges with fixed effects. In turn, the same specification adds price and freight charges to the disturbance if they are not fully captured by the fixed effects, for example, if the prices and freight charges are importer-product-origin-year specific. This is not a concern as long as our instruments based on port congestion and inspection are not systematically related to this information. To support this identification approach we provide test statistics and a series of robustness checks including lagging our instruments.

To facilitate estimation, given the substantial number of fixed effects, we estimate specification (8) in first differences. We cluster standard errors by the importing firm.

5.2. Import Regression Estimates

The left panel of Table 4 reports OLS (column 1) and 2SLS (column 2) estimates for the elasticity of imports with respect to processing time, $\gamma \chi$. Following equation (8), we relate log import values to the log of median processing times at the (importing)firm-product-origin-year unit of observation. We first difference our data. Therefore, the reported fixed effects are in addition to firm-product-origin fixed effects. First-stage statistics support

³⁸Heterogeneity across carriers accounts for variation in the ability of shipment handling in the supply chain, or, the possibility that more productive importers sort with more productive logistics providers.

the instrumental variable approach. For convenience, the right panel of Table 4 reports our quantification of import processing costs. We will discuss these estimates in the next section, but report them here because they derive from the elasticity estimates.

OLS estimates in Table 4 column 1 show that a one percent increase in the processing time reduces import values by 0.049 percent. Column 2 reports 2SLS estimates applying inspection probabilities and port congestion as instruments for endogenous processing times. The instrumental variable approach estimates an elasticity of negative 0.243 percent. The estimate is statistically significant at the 1 percent level. Therefore, as predicted in our theory, we conclude that longer border-processing times reduce import values.

As anticipated in the identification section, OLS estimates based on our proxy variable approach are subject to attenuation bias. The magnitude of the OLS estimates is about five times lower than the magnitude of the 2SLS estimate. Based on measurement error, the intuition is that fixed effects raise the noise to signal component of the identifying variation (Griliches and Hausman, 1986; McKinish, 2008). The observed bias compares to existing literature. For example, Costinot et al. (2012) and Paravisini et al. (2015) report similar magnitudes of attenuation bias.

Despite the increase in the magnitude of the IV estimate compared to OLS, the elasticity of import values with respect to import-processing time is still lower than estimates in the existing literature. Djankov et al. (2010) report that a 10 percent increase in the time it takes to move cargo from the factory gate to the ship reduces exports in the range of 4 percent. A possible explanation for the difference in elasticity estimates is that the existing elasticity estimates in the literature combine intensive and extensive margin variation since they are based on more aggregate data. However, in firm-level export data, Volpe Martincus et al. (2015) report that a 10 percent increase in customs export delays reduces exports by 3.8 percent.

We take away two insights from our results with respect to this existing literature. First, from an identification point of view, the magnitudes of our estimates are reasonable and comparable to the existing literature. Second, our elasticities based on import processing are lower than estimates in the literature based on export processing. Our theory provides an explanation. The elasticity $\gamma \chi$ increases in the shape of the processing-time distribution φ . An increase in φ results in a steeper processing distribution with a lower likelihood of long delays. Therefore, if import rules and regulations lead to a flatter processing distribution, a lower φ , then we would expect that imports are less elastic with respect to processing times.

First stage results provide evidence that our instruments work. As expected, congestion and a higher likelihood of inspection predict greater median processing times. The effective F statistics suggest that the instruments are not weak. Hansen's test statistic provides evidence that, after conditioning on fixed effects, overidentifying restrictions cannot be rejected.³⁹

As of now, we have estimates for the import elasticity, $\gamma \chi$, with respect to border processing times. We discuss the robustness of these elasticity estimates in section 7. Before doing that, the next section explains how we obtain the parameter estimates in the right hand panel of Table 4 to quantify border processing costs.

6. Border Processing Costs' Estimates

In the theory, the tariff equivalent processing cost equals $\lambda T^{\chi} - 1$ where $\chi = \varphi \vartheta / (\varphi + \vartheta)$ and λ is defined by equation (6). Estimating its magnitude requires information on $r/(\varphi - \omega)$ in addition to φ and ϑ . The right-hand panel of Table 4 reports structural parameters and processing costs that we explain in this section. We develop and discuss step-by-step structural estimates assuming an import demand elasticity of $\gamma = 4$ (column 3) consistent with the literature (Soderbery, 2015). For comparison, we also report structural estimates assuming an import demand elasticity of $\gamma = 6$ (column 4).

³⁹The tests for overidentifying restrictions is a test of joint-exogeneity and, as such, does not strictly provide information on the validity of the instruments, but on their coherence, i.e., whether they identify the same vector of parameters (see Parente and Santos Silva, 2012).

In order to evaluate the significance of these structural parameters and keeping things comparable, we bootstrap standard errors for all the structural parameters. We re-sample across firms and across strata when appropriate to account for clustering and estimate all parameters on these sub-samples of data. In all cases, the bootstrapped standard errors are based on 500 repetitions (see Efron and Tbshirani, 1994).

Dividing the import-processing-time elasticity from our IV estimation, 0.243, by the import demand elasticity, $\gamma = 4$, we obtain the processing-time-cost elasticity $\hat{\chi} = 0.061$. To put this parameter in perspective, Table 1 shows that at the median, document inspection more than doubles the processing time from 2 (Green) to 5 (Orange) processing days. Based on our elasticity of 0.061, automating physical document review to cut associated inspection times would reduce import processing costs by 5.4 percent. This implies a cost reduction of about 1.8 percent per day.

Next, we compute the elasticity ϑ using the processing-time cost elasticity $\hat{\chi} = \varphi \vartheta / (\varphi + \vartheta) = 0.061$. This requires an estimate for φ . In the Pareto distribution, the flatter is the shape determined by φ , the greater is the median processing time, T_{ihxy} , relative to the minimum processing time. Expressing the median as a function of the minimum processing time according to the Pareto distribution this means $T_{ihxy} = 2^{\frac{1}{\varphi}} \times t_{min,ihxy}$. Taking logs, we estimate the auxiliary regression $\ln T_{ihxy} = b_0 + b_1 \ln t_{min,ihxy} + u_{ihxy}$ and obtain $\hat{\varphi} = \frac{\ln(2)}{\hat{b}_0} = 2.072$. We estimate $b_1 = .87$ with standard error of 0.002. Applying $\hat{\varphi} = 2.072$ and $\hat{\chi} = 0.061$, we obtain $\hat{\vartheta} = \hat{\chi} \hat{\varphi} / (\hat{\chi} + \hat{\vartheta}) = 0.063$.

For intuition, when shipments do not get inspected, the average storage time is about 10 days. If we take this as our measure of lead time that firms allocate to clear the border, then an additional day of lead time increases lead-time costs by about .6 percent. This estimate is comparable to the low-end of time-cost elasticities estimated in Hummels and Schaur (2013).

As of now we focused on estimating the cost elasticity with respect to processing times and the associated percentage changes in processing costs. To determine the total tariff equivalent cost of import processing, we next quantify the level of cost by evaluating the cost multiplier λ .

Given our elasticity estimates, to compute λ we only need the ratio $r/(\varphi - \omega)$. Equation (4) says that all else equal, if $r/(\varphi - \omega)$ increases, then firms schedule greater lead times relative to processing times to avoid running late. Therefore, with observed information on processing times, elasticity parameters, and proxies for the lead time, we can estimate the constant $r/(\varphi - \omega)$ from equation (4).

To stay consistent with our identification strategy, we substitute median processing times for minimum processing times, $T = t_{min} \sqrt[q]{2}$, in equation (4) and rearrange to obtain

$$2\varphi^{-2}\vartheta T^{-\varphi}(t^*)^{\vartheta+\varphi} = \frac{r}{\varphi-\omega}.$$
(9)

Now, if we can generate an empirical counterpart to the left hand side of equation (9) in our data, then we can use it to estimate $r/(\varphi - \omega)$.

To generate the left hand side of equation (9) with our data, we substitute the parameters φ and ϑ with the estimates $\hat{\varphi} = 2.072$ and $\hat{\vartheta} = 0.063$. Next, as we did in (8), we substitute realized importer-product-origin-year specific observed median processing times, \hat{T}_{ihxy} , for the median processing time T. The only remaining variable to compute the left hand side of (9), is then the optimal lead time t^* .

Lead time measures the total time that firms optimally allocate to clear the border being mindful of delivery dates. It includes processing and storage steps. Firms' optimal lead times are not directly reported in our data. Instead, we approximate lead times with a measure based on the observable total border time. In our data, a shipment's total border time includes all storage and processing steps that the shipment follows in the port of entry. Then, as an approximation, we set the optimal unobserved lead time, t^* , equal to the median total border time of all shipments within each importer-productorigin-year observation. We define this importer-product-origin-year specific median of the total border time as \hat{t}_{ihxy}^* .

Substituting estimated parameters and observed measures of time into the left hand side of (9), and, considering that by doing so we measure the left hand side of (9) with random measurement error, e_{ihxy} , we obtain

$$2\hat{\varphi}^{-2}\hat{\vartheta}\hat{T}_{ihxy}^{-\hat{\varphi}}(\hat{t}_{ihxy}^{*})^{\hat{\vartheta}+\hat{\varphi}} = \frac{r}{\varphi-\omega} + e_{ihxy}.$$
(10)

Regressing the left hand side of (10) on a constant then returns the estimate $r/(\varphi - \omega) = 0.299$, as reported in column 3 of Table 4.

Applying all parameter estimates as reported in column 3 of Table 4 in equation (6), we then obtain $\hat{\lambda} = 1.104$. In our robustness section, we discuss several alternative choices for the lead time and examine λ 's sensitivity with respect to these alternative measures.

With all parameters at hand, we can now estimate the import cost due to border processing. Table 1 reports that across all shipments, the 50th percentile of the processing distribution is 3 days. Based on equation (5), we then compute the median tariff equivalent cost of import processing as $\tau^{\hat{Equiv}} = \hat{\lambda}T^{\hat{\chi}} = 1.104 \times 3^{0.061} = 1.181$. Therefore, evaluated at the 50th percentile, total border processing costs equal about 18 percent of the value of the import. The last line of Table 4 column 3 reports bootstrapped standard errors and shows that our estimate of the import processing tariff is statistically significant.

To evaluate the economic significance of the processing cost multiplier, set $\lambda = 1$. In this case the import processing cost simplifies to $T^{\hat{\chi}}$ and the import processing cost drops to about 7 percent, $3^{0.061} = 1.069$. Therefore, from an economic point of view, ignoring the multiplier would significantly under estimate border processing costs. Furthermore, even if a policy could reduce the median processing times to one day, then trade flows still experience a 10.4 percent import tariff due to costs of uncertainty captured by the multiplier.

Knowing the level of border processing costs, we can compare them to import tariffs

and potential tariff liberalizations. World Bank data show that the average world wide applied tariffs have decreased to about 6 percent in 2010.⁴⁰ Consequently, evaluated at the median, import processing costs are much greater than applied import tariffs. Eliminating all import tariffs, a reduction in tariffs of 100 percent, reduces import tariffs by 6 percentage points. If trade facilitation policy reduces processing times by 1.5 days⁴¹, then at the overall median processing time of 3 days, this reduces import processing costs by about 4.9 percentage points.⁴³

What would it take to achieve a 6 percentage point reduction in processing costs with trade facilitation policy? Table 1 shows that, at the 50th percentile, processing times in the orange channel equal 5 days, while processing in the green channel takes 2. Eliminating all document inspections, by switching from the orange channel to the green channel, therefore reduces processing times from 5 to 2 days at the median. This results in a 6.6 percentage point reduction in import processing cost.⁴⁴ Thus, according to our estimates, eliminating all document inspection, perhaps with the use of information technology, lowers import processing costs by the same amount as completely eliminating a 6 percent applied import tariff.

Following the same steps as above, we also estimate the model parameters at $\gamma = 6$. We report results in Table 4 column 4. In this case, a 10 percent increase in the processing time raises costs by $\chi = .4$ percent and the total border processing cost tariff equivalent drops to 11.3 percent. Based on existing demand elasticity estimates, we consider this value at the low end of the potential import processing tariff equivalent.

 $^{^{40}\}mathrm{See}$ http://data.worldbank.org/indicator/TM.TAX.MRCH.SM.AR.ZS 4142

⁴³This is a straight application of the tariff equivalent: $1.104 \times 3^{0.061} - 1.104 \times 1.5^{0.061} = 0.049$. ⁴⁴We compute $1.104 \times 5^{0.061} - 1.104 \times 2^{0.061} = 0.066$.

7. Robustness Checks

This section reports robustness checks for the import specification, the cost multiplier, and the processing cost. We start with the import specification and then explore the sensitivity of the cost multiplier.

7.1. Robustness Checks for Import Estimates

The following subsections examine the robustness of the instrumental variable estimates for the import regressions in the left hand panel of Table 4. We consider alternative definitions of the instrumental variables, import regulations and corruption, specification error, and aggregation bias.

7.1.1. Alternative Definitions of the Instruments and Learning

We start by examining the robustness of our main instrumental variable estimates, reported in Table 4, with respect to alternative definitions of the instruments. Results are reported in Table 5. Rows of results report various robustness checks. For all robustness checks, we re-estimate our import specification, equation (8), using 2SLS. In each case, we report the IV estimate for the effect of processing time on import values, $\gamma \chi$.

Before explaining details, it is straight forward to summarize the results. Across all robustness checks in Table 5, the estimated effect of the processing time on import values is very similar to our estimate in Table 4, -0.243.⁴⁵

The top panel of Table 5 examines robustness of our customs inspection instrument, channel. Our main specification uses the fraction of inspected shipments within importerproduct-origin-year observations as a measure of the probability of getting inspected. Volpe Martineus et al. (2015) propose an alternative instrument. Their instrumental variable is an indicator that equals one if more than 50 percent of the shipments in a given year within an existing trade relationship were inspected. The first row of Table

⁴⁵Our first-stage F-Statistics corroborate that our instruments are strong.

5 reports the results when we apply this median channel assignment as an instrument. The effect of processing times on imports remains similar as in the baseline IV estimate in Table 4.

The fraction of inspected shipments may be a noisy measure of the inspection probability in small samples. To examine this, we re-estimate our baseline focusing on annual observations for the instruments and all other variables that contain at least 10 and 20 transactions at the importer-product-origin-year level. Limiting the sample to annual firm-product-origin triplets that consist of at least 10 or 20 transactions also allows us to better assess whether the learning mechanism is at work, because such a sample includes observations with a sufficient number of shipments for firms to learn and update their prior about the median processing time to evaluate import processing costs as discussed in the identification section. Table 5 rows 2 and 3 show that over the samples that condition on at least 10 or 20 observations the effect of the border processing time on the import value remains negative and significant and similar to the baseline estimate, albeit slightly larger in magnitude.

The middle panel of Table 5 examines the robustness of our port congestion instrument. Firms' ability to update beliefs about the processing time may depend on the time window we consider before arrival of the shipment to compute the measure of congestion. For our main estimates, we focused on vessel arrivals the day before each shipment arrives at the port. We now extend that time window from 1 to 5 days. The estimates of border processing costs on import values are very similar to our baseline specification.

In the bottom panel of Table 5 we report estimates when we lag both instruments by one period. Even though our instruments are due to a random customs process and aggregate port congestion, one may be concerned that contemporaneous instruments are correlated with the contemporaneous disturbance. Coefficient estimates are similar to our main specification. A one log point increase in import processing time reduces import values by about 0.214 log points. For completeness, the last rows report results where we condition on observations that include at least 10 or 20 transactions in all variables. The elasticity remains negative and significant. The magnitudes of the elasticities decrease slightly compared to the baseline estimates. A potential reason is that these esimates emphasize lagged information as predictors of processing time and de-emphasize the within-year learning mechanism.

7.1.2. Alternative Specifications of Fixed Effects

In Table A5 we examine robustness of the baseline IV estimate reported in Table 4 with respect to alternative specifications of fixed effects. We estimate the import regression, equation (8), with 2SLS applying our standard inspection and port congestion instrument as explained in the identification section, but vary the set of fixed effects. More rigorous fixed effects lend credibility that our instruments meet the exclusion restriction, but they also absorb useful identifying variation.

The first row of results in Table A5 reports the IV estimates for the effect of processing time in import values. Across the columns, the effect varies between -0.19 and -0.268. This is remarkably similar to the effect we report in Table 4, -0.243. First stage results show that both the congestion instrument and the inspection instrument significantly predict the processing time. F-statistics confirm the strength of the instruments. We conclude that the choice of fixed effects does not significantly affect our results.

Finally, even if we account for Origin-Product-Year, or, Firm-Origin-Year, or, Firm-Product-Year fixed effects, there is still sufficient variation in the instruments to predict the first stage. Consequently, we conclude that there is relevant importer-product-origin variation within years to identify the coefficients.

7.1.3. Import Processing Regulations, Tariffs, and Non-Tariff Measures

In Table 6 we examine how regulations in import processing, tariffs, and non-tariff measures affect our conclusions. Across the columns, we report 2SLS estimates for the effect of the log processing time on log import values according to equation (8). The instruments are port congestion and inspection frequencies as explained in the identification section. Across the columns, we estimate the effect from various sub-samples that exclude shipments subject to special regulations and policies.

To clear the border Peruvian firms may use an express channel for their shipments. This channel allows firms to file customs documents while still in transit. Column 1 provides estimates when we drop shipments that cleared through this channel. The coefficient estimate on the processing time equals -0.247, is statistically significant, and comparable to the baseline estimate in Table 4, -0.243.

In column 2 we report estimates focusing on low tariff products. We consider trade flows with less than a 5% tariff. Dutt and Traca (2010) and Sequeira (2016) consider the possibility of tariff evasion. The concern is that especially when tariffs are high firms interact with officials to lower their tariff burden. In that case, they may also attempt to reduce the processing burden. The estimates remain comparable to our baseline.⁴⁶ We conclude that tariff evasion does not significantly affect our conclusions.

Column 3 augments the baseline specification with ad-valorem freight, tariff, and insurance charges.⁴⁷ The theory maintains that tariffs, transportation, and insurance markets are independent of processing time. If transportation providers' quality depends on processing speed and higher quality providers charge higher rates, then excluding this information may result in omitted variable bias. Our results show that this is not the case. Estimates remain similar to the baseline estimates. We conclude that freight charges do not lead to omitted variable bias and that our fixed effects sufficiently account for this information.

Column 4 reports results where we exclude all products that require special import permits (Bown and Crowley, 2016; Carballo et al. 2016b). These products require additional processing that may affect the estimates. Again, we conclude that this does

⁴⁶We also estimated the model for high tariff products and the estimates are comparable.

⁴⁷We first compute τ_{ihxy} as the sum of f.o.b import value, freight charges, insurance charges, tariff charges and divide this sum by the f.o.b. value. We then augment our baseline specification with τ_{ihxy} .

not affect our baseline estimates and that our fixed effects sufficiently account for this potential product heterogeneity.

7.1.4. Aggregation

Next we examine a seeming disconnect between the theory and the empirics. In the theory, we model the processing costs for an individual shipment. In the data, as is standard in many empirical trade papers involving firm level data, we aggregate to annual levels. There are several reason why some level of aggregation is useful. It keeps the results comparable to the literature. Also, at the transaction level trade data includes tiny one-time shipments due to experimentation or emergency shipments unrelated to our theory. On the other hand, aggregating to annual levels potentially eliminates useful identifying variation within annual observations. Table 7 reports two robustness checks with respect to this aggregation problem.

First, we estimate the effect of import processing times on import values according to specification (8) with quarterly data and monthly data. To do so, we construct the import processing time and associated congestion and inspection instruments at the quarterly and monthly levels. Quarter-year and month-year fixed as well as country-product-quarter and country-product-month fixed effects account for product characteristics as well as seasonality and shopping seasons. Columns 1 and 2 of Table 7 report the 2SLS estimates for quarterly and monthly data respectively including quarter-year and month-year fixed effects. A one percent increase in processing time reduces imports by about 0.154 percent at quarterly observations and 0.130 percent at monthly observations. Columns 3 and 4 report similar results for specifications when we account for country-product-quarter and country-product-month fixed effects. Overeall, the estimates are slightly smaller than the baseline estimate in Table 4. Perhaps this is to be expected, as the coefficient captures the effect of the processing time only for a quarter as opposed to the entire year. Furthermore, quarterly and monthly observation limit identifying variation due to learning mechanisms as explained in Section 5.1.

Second, we examine the effect of import processing time on the import value per shipment. To do so, we divide the annual import value by the total number of shipments within each firm-product-origin-year observation and take logs. We then estimate the effect of log processing times on the log import value per shipment applying 2SLS and our congestion and inspection instruments. Table 7 column 5 reports the results. The effect of the processing time on the import value per shipment is -0.238 and almost identical to the effect of processing times on total import values reported in Table 4. We conclude that considering the number of shipments does not affect the results.⁴⁸

Finally, we estimate specification (8) at the (importing)firm-product-carrier-exporter(firm)year unit of observation. We re-construct our instruments and estimate 2SLS. We extend the fixed effects to account for exporting firm heterogeneity. In addition, we account for heterogeneity across carriers.⁴⁹ Ben-Daya and Abdul (1994) consider that firms may shorten lead times, but at an added cost. A way to accomplish this may be to choose faster carriers. We note that our identification approach relies on processing times, not lead times. Nevertheless, accounting for carrier fixed effects accounts for this mechanism. Appendix Table A6 reports the results. Across all specifications an increase in import processing times reduces imports. Coefficient estimates are comparable to the baseline IV estimate in Table 4.

7.1.5. Alternative Measures of Border Time

In the previous sections we estimate import-processing cost elasticities based on the actual processing time as defined in Section 2. In this subsection, we examine if the definition of the time it takes to import matters for elasticity estimates.

 $^{^{48}}$ For theory that determines shipping frequency see Hornok and Koren (2015a, 2015b) and Kropf and Sauré, (2014).

⁴⁹Unfortunately we do not observe customs brokers in our data. If carriers sort systematically across customs workers, then we expect that carriers fixed effects account for this heterogeneity.

Measurements of the time it takes to import vary across publicly available data sources. For example, in 2017, the Enterprise Survey reports the number of days to clear shipments from customs in Peru as 14 days, on average.⁵⁰ Compared to our statistics in Section 2, this measure is closer to our definition of the total time to clear imports, including storage steps in the import process, rather than actual processing time. For comparison, the 2010 Doing Business Business Trading Across Border's data reports that the time to import into Peru is 24 days. However, a recent methodology change results in much lower measures, 72 hours for border compliance and 72 hours for documentary compliance to cross the border.⁵¹ Thus, while the previous methodology seems more consistent with our definition of the total time shipments take to import, the current methodology is closer to our measure of processing time. Does the distinction between processing time and total border time matter for elasticity estimates? We use our data to answer this question.

We estimate our import specification, equation (8), but instead of our measure of processing time, we focus on a measure of median total border time. As defined in Section 6, let \hat{t}_{ihxy}^* be the median total border time (including all storage steps in the import process) of all shipments within each importer-product-origin-year observation. Then, we estimate

$$ln(v_{ihxy}) = \delta_{hxy} + \kappa_{iy} + \kappa_{ihx} + \beta ln \hat{t}^*_{ihxy} + u_{ihxy}.$$
(11)

Table 8 reports OLS and 2SLS estimates. The instruments are inspection rates and port congestion, as explained in the identification section. The estimated import elasticities with respect to the total border time are -0.057 and -0.556. For comparison, Table 4 reports import elasticity estimates with respect to the median processing time as -0.049 and -0.243 for OLS and IV estimates. Therefore, our preferred import elasticities

⁵⁰https://www.enterprisesurveys.org/en/data/exploreeconomies/2017/peru#trade

⁵¹http://www.doingbusiness.org/data/exploretopics/trading-across-borders/whatmeasured

based on processing time lead to a more conservative estimate of the import processing cost elasticity.

7.2. Robustness Checks for Cost Multipliers

In addition to elasticity parameters, the key component we back out from the data to obtain λ is $r/(\varphi - \omega)$. In section 6 we used a measure of total border time, \hat{t}_{ihxy}^* , as proxy for lead time. We examine the sensitivity of λ with respect to several alternative choices of lead time proxies.

Equation (9) shows that our approach overestimates $r/(\varphi - \omega)$ and therefore λ , if our lead time proxy is greater than the optimally chosen lead time. All else equal, the greater the proxy for t^* , the greater $r/(\varphi - \omega)$.

Our lead time proxy overstates the unobserved lead time chosen by firms if, for example, a firm chooses a lead time of five days, but the actual processing takes six. In this case, we observe a total border time of six days and overstate the optimal lead time by one day. Then, the identification issue is that the longest total border times are actually measures of long processing time instead of lead time. To examine this, we use our transaction level data. First, we compute the median processing time at firm-product-origin-year observations. Second, we drop the highest 5th and 10th percentiles of total border time. We then compute the median total border time, \hat{t}^*_{ihxy} , over this more limited sample as proxy for our lead time measure. Following the same approach as in section ??, but using the corrected lead time proxy, we obtain a new $\hat{\lambda}$.

Table 9 column (1) repeats the baseline estimates for comparison. Columns (2) and (3) show the results with the corrected measures with different cutoffs. We find that by correcting the lead time proxy $r/(\varphi-\omega)$, λ increase compared to the baseline specification. This means that when total border times are high in our sample, they are high due to long storage times instead of unusually high processing times.

Next, we recognize that we do not observe the ocean transit time. It is possible that

the storage time we observe at the border is not just buffering for random shocks in border processing, but also captures the lead time for ocean transit. To examine the sensitivity of λ with respect to this data problem, we make two adjustments.

First, we focus on countries that are close by to eliminate lengthy ocean transit times.⁵² In this case, we re-estimate all of the structural parameters over the restricted sample and compute $\hat{\lambda}$. Table 9 column (4) shows the results. The multiplier λ increases due to a greater cost of late delivery, $r/(\varphi - \omega)$. Therefore, imports sourced from countries close by are subject to especially high costs of running late. This evidence complements Evans and Harrigan (2003) who provide evidence that firms move closer to the destination market if they face short selling seasons and high demand uncertainty.

Second, we focus on the top 6 source countries in the sample.⁵³ We collect average ocean transit times from searates.com and add them to the processing time and total border time for that sample. Then we re-estiamte the elasticities and the cost parameters. Table 9 column (5) shows the results. With increased time measures due to ocean transit, λ decreases by about 4 percentage points. Therefore, accounting for ocean transit time results in slightly lower import processing costs.

Finally, it is possible that shipments are stored after clearing customs, which we do not observe in our data. If there is storage after the port then our lead time proxy, \hat{t}_{ihxy}^* , underestimates optimal lead time, t^* , and we underestimate λ . In this case, our cost multipliers are conservative estimates for import processing costs.

8. Border Processing Costs, New Trade Relationships, and Product Categories

In this section we examine if import processing-costs differ across trade relationships and different product categories.

⁵²More specifically, we consider Ecuador, Chile, Colombia, Panama, Costa Rica, Nicaragua, Guatemala, Mexico, Brazil, Argentina and Uruguay as the closest countries.

⁵³More specifically, we consider import flows from China, United States, Germany, Italy, Spain, and Brazil.

8.1. Estimates by New versus Experienced Importers

Recent literature examines the importance of new trade relationships (Bernard et al., 2017a, 2017b). The hope for trade facilitation policy often is that it especially reduces trade costs and affects export growth dynamics by reducing costs for new importers, exporters, and trade relationships.⁵⁴ Less experienced firms may find it more challenging to comply with regulations and experience greater import processing costs. On the other hand, more experienced firms may run more complicated supply chains. In that case, we expect that processing delays are especially costly. Therefore, how firms' import experience relates to border processing costs is an empirical question. We provide evidence for this in Table 10. Column 1 reports estimates for new importers, firms that never imported before. Column 2 reports estimates for experienced importers. To obtain these estimates, we run one regression, but interact the processing time and the instruments with an indicator to distinguish new from old importers.

Estimates show that, as a tariff equivalent, import processing costs for new importers are more than double the cost of experienced importers, 28.3 percent of the import value versus 11.8 percent. This difference is driven by both cost parameters and longer processing times. At the median, shipments by new importers take eight days to clear while shipments of established importers take only four. Therefore, trade facilitation policy that lowers median processing times for new importers can be especially effective in lowering import costs to increase trade. However, the difference in costs between new and experienced importers is not only due to differences in processing times. Processing costs of new importers are subject to a greater cost elasticity (0.067 versus 0.041) and multiplier (1.117 versus 1.057). The difference in the multiplier derives from the fact that new importers have greater costs of running late as captured by $r/(\varphi - \omega)$. Therefore, the differences in costs between new and experienced importers are not only due to differences

⁵⁴https://www.wto.org/english/news_e/news17_e/fac_31jan17_e.htm

in processing times, but also due to supply chain management strategies as captured by the cost parameters.

8.2. Estimates by Product Categories

Using End-Use categories, Hummels and Schaur (2013) report estimates of time-cost elasticities for five broad product groups including Foods and Beverages, Industrial Supplies, Capital Goods, Automotive, and Consumer Goods. We employ Broad Economic Categories and report total import processing costs for the same product groups in Table 11. The theory shows that magnitudes of time cost elasticities are only comparable if the shape of the processing distribution is the same. However, if the shape of the processing distribution is relatively similar across products, then we may focus on relative comparisons.

We find that processing-time cost elasticities, χ , and lead-time cost elasticities, ϑ , are low for Food and Beverages and Transport Equipment, and relatively high for Industrial Supplies, Consumer Goods, and Capital Goods. This pattern is broadly consistent with the literature.⁵⁵. Inputs, captured by Industrial Supplies, and consumer goods, particularly if including fashion items (Evans and Harrigan, 2005) are time- sensitive.⁵⁶

While consistent with the literature, it is surprising that Foods and Beverages, including perishable items, have a relatively low processing-time cost elasticity, χ , compared to Capital Goods. However, Foods and Beverages have a greater cost multiplier λ . Therefore, adjusting for costs associated with uncertainty using the multiplier is relevant to determine the time-cost rankings of these two product categories. Taking into account the multiplier and the processing time cost elasticity, Industrial Supplies and Consumer

 $^{^{55}\}mathrm{See}$ Hummels and Schaur (2013), Table 7 Columns 6 to 10

⁵⁶A notable difference is that while we find a relatively low time-cost elasticity for transportation equipment, Hummels and Schaur (2013) report a relatively high elasticity for that sector. There are several potential reasons, including that the relatively high time-cost elasticity for that product group in Hummels and Schaur (2013) is due to an unusually low substitution elasticity, which amplifies the time-cost elasticity. They estimate $\sigma = 0.9$.

Goods experience the greatest import-processing costs. Consumer Goods in particular experience a large costs associated with uncertainty $(\lambda - 1 = 0.131)$.

The question arises if import patterns in Peru, and specifically the Port of Callao, are comparable to import patterns in other countries like the U.S. Appendix Figure A.1 shows the product mix of imports in the U.S., based on WITS data, and import shares of the Port of Callao based on out trade data. Import shares across broad products for Peru and the US are surprisingly similar. The main exceptions are that Callao imports less mineral products and vehicles including vessels and aircraft, but more plastics. Across the other products, there are differences in magnitudes, but Callao shares tend to be high when the U.S. shares tend to be high as well.

9. Conclusions

Trade facilitation is a major policy initiative that means to lower trade costs by reducing regulatory burden to accelerate international supply chains. However, policies related to non-tariff barriers are difficult to measure and evaluate (Goldberg and Pavcnik, 2016). As a result, the level of trade processing cost and the potential for trade facilitation policy is not clear. To make progress, this paper focuses on a common measure of non-tariff import costs, the time it takes to process shipments at the border. We use theory to translate processing times into costs. Then, we employ highly detailed import data to estimate these costs.

We draw several policy relevant conclusions. Costs associated with import processing are high and trade facilitation policies to reduce median processing times have the potential to substantially reduce these costs. For example, we provide evidence that a policy that eliminates delays due to document inspection could reduce border processing costs by about six percentage points. This is comparable to eliminating the average worldwide applied tariff of about six percent.⁵⁷

⁵⁷https://data.worldbank.org/indicator/TM.TAX.MRCH.WM.AR.ZS?

Our evidence also provides insights for trade theory. Import processing costs are especially high for new importers. Therefore, trade theory that formalizes trade policy to reduce non-tariff barriers associated with border processing costs ought to consider the formation of new trade relationships.

We apply our theory and identification strategy to Peru due to data availability. However, delays in the import process and concerns of port efficiency are not unique to Peru (Blonigen and Wilson, 2008). Recently, long processing times in the port of Los Angeles have received much attention.⁵⁸ This raises the question of external validity of our parameter estimates to evaluate, for example, import processing times at the port of LA. Our theory provides fundamentals for our parameter estimates. As a consequence, our approach is clear under what pooling restrictions our results apply to other countries and links of the international supply chain.

As more detailed customs data sets with detailed information on processing times become available, the question of how firms price delays, i.e., what do they know about delays when they make their import or export decisions, will be important for identification strategies. We expect that this is a fruitful future area of research to determine how firms optimally respond to trade barriers and if the benefits of policies such as the Agreement on Trade Facilitation justify their costs.

⁵⁸https://www.marketplace.org/2021/03/08/dozens-container-ships-waiting-unloaded-port-los-angeles/

References

- [1] Antràs, Pol; Teresa C. Fort, and Felix Tintelnot, 2017. The Margins of Global Sourcing: Theory and Evidence from U.S. Firms, American Economic Review, 107, 9.
- [2] Ben-Daya M. and Raouf Abdul. 1994. Inventory Models Involving Lead Time as a Decision Variable, The Journal of the Operational Research Society, Vol. 45, No. 5, pp. 579-582.
- [3] Bernard, Andrew B.; Moxnes Andreas; Ulltveit-Moe Karen Helene, 2017a. Two-Sided Heterogeneity and Trade. Working Paper.
- [4] Bernard, Andrew B.; Moxnes Andreas; Saito Yukiko U., 2017b. Production Networks, Geography and Firm Performance. Working Paper.
- [5] Blonigen, B. and Wilson, W., 2008. Port efficiency and trade flows. Review of International Economics, 16.
- [6] Boehm, Christoph E , Aaron Flaaen , and Nitya Pandalai-Nayar (2019): Input Linkages and the Transmission of Shocks: Firm-Level Evidence from the 2011 Tohoku Earthquake. The Review of Economics and Statistics, 101:1, 60-75
- [7] Bown, C.P. and Crowley M.A., 2016. The empirical landscape of trade policy, Handbook of Commercial Policy, 1(A):3-108, Edited by Kyle Bagwell and Robert Staiger.
- [8] Carballo, J.; Graziano, A.; Schaur G.; and Volpe Martincus, C., 2016. Endogenous Border Times. IDB Working Paper Series Nr: IDB-WP-702.
- [9] Carballo, J.; Graziano, A.; Schaur G.; and Volpe Martincus, C., 2016b. The Border Labyrinth: Information Technologies and Trade in the Presence of Multiple Agencies. IDB Working Paper Series Nr: IDB-WP-706.
- [10] Costinot Arnaud, Donaldson Dave and Komunjer Ivana, 2012. What Goods do Countries Trade? A Quantitative Exploration of Ricardo's Ideas. Review of Economic Studies, 79, 581-608.
- [11] Djankov, S.; Freund, C.; and Pham, C., 2010. Trading on time. Review of Economics and Statistics, 92, 1.
- [12] Dutt P and Traca D. 2010. Corruption and bilateral trade flows: extortion or evasion? Review of Economics and Statistics, 92, 4, 843-860.
- [13] Efron, Bradley, and Robert J. Tibshirani. An Introduction to the Bootstrap. CRC press, 1994.
- [14] Evans, Carolyn L. and Harrigan James, 2005. Distance, time and specialization: Lean retailing in general equilibrium. American Economic Review, 95(1):292-313.
- [15] Fernandes, Ana Margarida, Hillberry, Russell Henry, Mendoza Alcantara, Alejandra, 2021. Trade effects of customs reform: evidence from Albania. The World Bank Economic Review, Volume 35, Issue 1, February 2021, Pages 34-57.
- [16] Fitzgerald, Doireann, Haller Stefanie and Yedid-Levi Yaniv, 2017, How exporters grow, NBER w21935.
- [17] Freund, Caroline, and Rocha, Nadia. 2011. What Constrains Africa's Exports?, World Bank Economic Review 25(3): 361-386.
- [18] Goldberg P. K. and Pavcnik N., 2016, The Effects of Trade Policy, NBER 21957.
- [19] Gopinath, Gita, and Brent Neiman. 2014. Trade Adjustment and Productivity in Large Crises. American Economic Review, 104(3): 793-831.
- [20] Griliches, Z., and J. A. Hausman. 1986. Errors in Variables in Panel Data. Journal of Econometrics 31:93-118.
- [21] Halpern, László, Miklós Koren, and Adam Szeidl. 2015. Imported Inputs and Productivity. American Economic Review, 105(12): 3660-3703.
- [22] Heid B, Larch M, and Yotov Y., 2017. Estimating the Effects of Non-discriminatory Trade Policies within Structural Gravity Models, School of Economics Working Paper Series 2017-10, LeBow College of Business, Drexel University.
- [23] Hornok, C. and Koren, M., 2015a. Administrative barriers to trade. Journal of International Economics, 96.
- [24] Hornok, C. and Koren, M., 2015b. Per-shipment costs and the lumpiness of international trade. Review of Economics and Statistics, 97, 2.

- [25] Hummels, David L., and Georg Schaur. 2013. Time as a Trade Barrier. American Economic Review, 103 (7): 2935-59.
- [26] Kropf, A. and Sauré, P., 2014. Fixed cost per shipment. Journal of International Economics, 92, 1.
- [27] McKinish, T., 2008. Panel Data Models and Transitory Fluctuations in the Explanatory Variable In Modeling and Evaluating Treatment Effects in Econometrics, eds. Daniel L. Millimet, Jeffrey A. Smith, and Edward J. Vytlacil, 335–58. Amsterdam: Elsevier
- [28] Melitz, M. J., 2003. The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity. Econometrica. 71(6), 1695-1725
- [29] Oberhofer, Harald and Pfaffermayr, Michael and Sellner, Richard, 2018. Revisiting Time as a Trade Barrier: Evidence from a Panel Structural Gravity Model CESifo Working Paper No. 7426
- [30] Alfonso Irarrazabal, Andreas Moxnes and Luca David Opromolla, 2015. The Tip of the Iceberg: A Quantitative Framework for Estimating Trade Costs, The Review of Economics and Statistics, vol. 97(4), pages 777-792,
- [31] Paravisini Daniel, Rappoport Veronica, Wolfenzon Daniel, Schnabl Philipp, 2015. Dissecting the Effect of Credit Supply on Trade: Evidence from Matched Credit-Export Data. The Review of Economic Studies, 82(1), 333-359
- [32] Parente, P., Santos Silva, J., 2012. A Cautionary Note Tests of Overideintifying Restrictions. Economic Letters, 115.
- [33] Persson Maria, 2008. Trade Facilitation and the EU-ACP Economic Partnership Agreements. Journal of Economic Integration. 23, 518-546.
- [34] Rodrigue Joel and Yong Tan, 2019. Price, Product Quality and Exporter Dynamics: Evidence from China, working paper
- [35] Sequeira, S., 2016. Corruption, Trade Costs, and Gains from Tariff Liberalization: Evidence from Southern Africa. American Economic Review, 106, 10.
- [36] Soderbery, A., 2015. Estimating import supply and demand elasticities: Analysis and implications. Journal of International Economics, 96, 1.
- [37] SUNAT, 2010a. INTA-PG.01. Importacion para el consumo: Procedimiento general. Lima, Peru.
- [38] SUNAT, 2010b. INTA-PE.00.03. Procedimiento especifico: Reconocimiento físico, extraccion y analisis de muestras. Lima, Peru.
- [39] Volpe Martincus, C.; Carballo, J.; and Graziano, A., 2015. Customs. Journal of International Economics, 96, 1.

10. Tables

			Doroontilo						
Stage	Channel	Average	F . 1	10/1	0511	Foul		00/1	05/1
		_	5th	10th	25th	50th	75th	90th	95th
Total Border Time	All	14.7	4	5	7	10	18	29	42
	Green	11.6	4	4	6	8	13	21	31
	Orange	15.7	5	6	8	13	20	28	38
	Red	23	7	8	13	19	28	42	56
Processing Time	All	4.3	1	2	2	3	6	12	16
	Green	1.8	1	1	2	2	3	5	6
	Orange	6	2	2	3	5	9	13	17
	Red	10.3	3	4	6	9	15	20	24
Storage Time	All	10.4	2	3	4	$\overline{7}$	12	20	31
	Green	9.8	2	3	4	6	11	19	28
	Orange	9.8	2	3	5	7	12	20	28
	Red	12.7	2	3	5	8	15	27	40

Table 1: Border Times (in Days): Total and Stages in 2013, by Customs Verification Channel

Source: Authors' calculations based on data from SUNAT.

The table reports the average and percentiles of the distribution of the total time to import, the total processing time, and storage time by customs verification channel (i.e., green, orange, and red) for 2013. The sample corresponds to all maritime imports entering into Peru through the port of Callao.

	(1)	(2)	(3)	(4)
Unloading Time	-0.152^{***}	-0.169^{***}	-0.111***	-0.132***
	(0.011)	(0.013)	(0.011)	(0.012)
Firm Fixed Effect	Yes	No	Yes	No
Product-Origin Fixed Effect	Yes	No	Yes	No
Firm-Product-Origin Fixed Effect	No	Yes	No	Yes
Day Fixed Effects	No	No	Yes	Yes

Table 2: Effect of Unloading Time on Storage Time

The table presents the effect of unloading times on storage times conditional on fixed effects. The dependent variable is the natural log of storage time and the main explanatory variable is the natural log of the unloading time at the port. Standard errors clustered at importing firm-level are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

	All Imports							
Year Imp	Import Value	Number of	Number of	Number of				
	import value	Importers	Origins	Products				
2007	19,100	19,290	199	6,989				
2008	27,900	$22,\!542$	205	$6,\!230$				
2009	$20,\!600$	$23,\!597$	201	$6,\!174$				
2010	28,200	$25,\!592$	203	6,233				
2011	36,100	$26,\!804$	210	$6,\!177$				
2012	40,200	28,799	211	6,302				
2013	41,100	$30,\!131$	209	6,303				
	Pe	rcentage Share	Callao					
2007	72.3	64.0	86.4	92.4				
2008	72.4	65.4	87.3	92.6				
2009	73.8	65.7	93.0	93.0				
2010	75.5	64.8	84.7	92.9				
2011	76.7	65.8	84.8	93.2				
2012	75.9	65.5	90.5	93.3				
2013	74.7	65.6	88.5	93.2				

Table 3: Aggregate Import Indicators

The table reports aggregate import indicators for each year of our sample period. In the first panel, all imports are considered. Import values are expressed in millions of US dollars. In the second panel, only maritime imports entering through Callao are considered. This panel shows the percentage share of total Peruvian imports accounted for by these maritime imports along the dimensions that correspond to the selected indicators.

Estimation			Qu	antification	l
	(1)	(2)		(3)	(4)
	OLS	IV		$\gamma=4$	$\gamma=6$
Processing Time $(\gamma \chi)$	-0.049^{***} (0.005)	-0.243^{***} (0.015)	Χ	$\begin{array}{c} 0.061^{***} \\ (0.004) \end{array}$	0.040^{**} (0.004)
Congestion		First Stage 0.028***	arphi	2.072^{***} (0.037)	2.072^{**} (0.037)
Channel		(0.000) 0.743^{***} (0.009)	θ	$\begin{array}{c} 0.063^{***} \\ (0.007) \end{array}$	0.041^{**} (0.007)
F-Test		4,317.239 [0.000]	$r/(arphi-\omega)$	0.299^{***} (0.039)	0.189^{**} (0.039)
Hansen Test		$\begin{bmatrix} 0.025 \\ [0.874] \end{bmatrix}$	$(\lambda - 1)$	0.104^{***} (0.008)	0.066***
Fixed Effect					
Firm-Year	Yes	Yes	$(\lambda \cdot T^{\chi} - 1)$	0.180^{***}	0.113**
Origin-Product-Year	Yes	Yes		(0.012)	(0.011)
Observations	589,842	589,842			

Table 4: Effect of Processing Time on Imports and Processing Costs

The table reports OLS and IV estimates of equation (8) along with the first stage estimates and the effective F-test statistics and the Hansen test statistics for the latter. The dependent variable is the change in the natural log of import values at the firm-product-origin-year level. In the IV estimation, the instruments are port congestion as proxied by the median number vessels that arrived at the port the day before the vessel carrying the shipment in a given year, and the average allocation to inspection (either documentary or physical) in a given year. Firm-year and origin country-product-year fixed effects are included (not reported). Standard errors clustered by importing firm are reported in parentheses below the estimated coefficients. Unit of observation: importing firm by origin by product by year. In the case of the right panel (Quantification), bootstrapped standard errors with 500 replications are reported. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

	IV Estimate				
	Processing Time				
Robustness Channel Instrument					
Median Channel	-0.239***				
	(0.014)				
10 or more transactions	-0.285***				
	(0.029)				
20 or more transactions	-0.295^{***}				
	(0.039)				
Robustness Congestion Instrument					
Window: 2 Days	-0.238***				
	(0.015)				
Window: 3 Days	-0.239***				
	(0.015)				
Window: 4 Days	-0.238***				
	(0.015)				
Window: 5 Days	-0.239***				
	(0.015)				
Robustness Lagged Instruments					
Lag 1	-0.214***				
	(0.013)				
Lag 1 and 10 or more transactions	-0.153**				
	(0.076)				
Lag 1 and 20 or more transactions	-0.176**				
	(0.087)				

Table 5: Robustness Checks: Instrumental Variables

Source: Authors' calculations based on data from SUNAT.

The table reports IV estimates of equation (8). The dependent variable is the change in the natural log of the import value at firm-product-originyear level. The independent variable is the change in the log of the import processing time. Firm-year and origin country-product-year fixed effects are included (not reported). Unit of observation: importing firm by origin by product by year. Standard errors clustered by firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

	(1)	(2)	(3)	(4)
	No Express	Low Tariffs	Transport Quality	No Permits
Processing Time	-0.247***	-0.235***	-0.240***	-0.241***
	(0.016)	(0.020)	(0.015)	(0.016)
Trade Costs			-1.535***	
			(0.068)	
		First S	Stage	
Congestion	0.028***	0.029***	0.028***	0.029***
	(0.001)	(0.001)	(0.001)	(0.001)
Channel	0.744^{***}	0.719^{***}	0.742^{***}	0.733^{***}
	(0.009)	(0.008)	(0.008)	(0.009)
F-Test	4,249.0	3,705.1	$4,\!317.0$	3,727.2
	[0.000]	[0.000]	[0.000]	[0.000]
Fixed Effect				
Firm-Year	Yes	Yes	Yes	Yes
Origin-Product-Year	Yes	Yes	Yes	Yes
Observations	566,082	343,002	589,842	493,384

Table 6: Robustness Checks: Port and Customs Regulation

The table reports IV estimates of equation (8) along with the first stage estimates and the effective F-test statistics. The dependent variable is the change in the natural log of the import value at the firm-product-origin-year level. The main explanatory variable is the change in the natural log of the median processing time. The instruments are inspection frequency and port congestion. In column (1) imports processed through the expressed channel are excluded. In column (2) imports with tariffs above 5% are excluded. In column (3), the baseline regression is augmented incorporating the change in the natural log of the freight, tariff and insurance costs at firm-product-origin-year. In column (4) imports from products with additional documents required are excluded. Standard errors clustered by firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

Table 7: Robustness	Checks:	Aggregation
---------------------	---------	-------------

	(1)	(2)	(0)	(4)	(2)
	(1)	(2)	(3)	(4)	(5)
	Quarterly	Monthly	Quarterly	Monthly	Value Per
					Shipment
Processing Time	-0.154***	-0.130***	-0.160***	-0.137***	-0.238***
	(0.013)	(0.012)	(0.013)	(0.013)	(0.014)
	Fi	rst Stage			
Congestion	0.024***	0.087***	0.024***	0.086***	0.028***
	(0.001)	(0.003)	(0.001)	(0.003)	(0.001)
Channel	0.715***	0.693***	0.715***	0.694***	0.743***
	(0.008)	(0.009)	(0.009)	(0.009)	(0.009)
F-Test	4737	4368	4847	4201	4317.239
	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]
Fixed Effects:					
Firm-Year	Yes	Yes	Yes	Yes	Yes
Country-Product-Year	Yes	Yes	Yes	Yes	Yes
Country-Product-Frequency	No	No	Yes	Yes	Yes
Frequency-Year	Yes	Yes	Yes	Yes	No
Observations	2,020,086	$2,\!676,\!416$	1,967,939	2,523,212	589,842

The table reports IV estimates of alternative specifications of equation (8) along with the first stage estimates and the F-test statistics. For quarterly estimates the dependent variable is the change in the natural log of the import value at the importing firm-product-origin-quarter-year level. For monthly estimates the dependent variable is the change in the natural log of the import value at the importing firm-product-origin-month-year level. For value per shipment estimates the dependent variable is the log changes in the log annual value per shipments. Standard errors clustered by firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; ***

	(1)	(2)
	OLS	IV
Total Time	-0.057***	-0.556***
	(0.005)	(0.026)
	First	t Stage
Congestion		0.009***
		(0.000)
Channel		0.281^{***}
		(0.003)
F-Test		834
		[0.000]
Fixed Effects:		
Firm-Year	Yes	Yes
Origin-Product-Year	Yes	Yes
Observations	589,842	589,842

Table 8: Robustness Checks: Effect of total border time on Imports

Source: Authors' calculations based on data from SUNAT.

The table reports OLS and IV estimates of log import values on the log of the total border time along with the first stage estimates and the F-test statistics for the latter. Standard errors clustered by firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

	(1)	(2)	(3)	(4)	(5)
	Baseline	Lead	Time	So	urcing
		5%	10%	Close	Ocean Time
χ	0.061***			0.073***	0.066***
	(0.004)			(0.003)	(0.004)
φ	2.072^{***}			1.888^{***}	1.958^{***}
	(0.037)			(0.039)	(0.036)
θ	0.063^{***}			0.077^{***}	0.069^{***}
	(0.007)			(0.004)	(0.005)
$r/(arphi-\omega)$	0.299***	0.600***	0.607***	0.615***	0.078***
	(0.039)	(0.075)	(0.076)	(0.0789)	(0.049)
$(\lambda - 1)$	0.104^{***}	0.127^{***}	0.128^{***}	0.155^{***}	0.0631^{***}
	(0.008)	(0.009)	(0.009)	(0.008)	(0.007)
$(\lambda \cdot T^{\chi} - 1)$	0.180^{***}	0.205^{***}	0.206^{***}	0.235^{***}	0.136^{***}
	(0.012)	(0.015)	(0.014)	(0.010)	(0.012)

Table 9: Robustness Checks: Different Lead Time Measures and Sourcing Patterns

Column (1) reports our baseline estimates. Columns (2) and (3) re-estimate $r/(\varphi - \omega)$ and subsequent parameters dropping all the observations where the difference between total time and processing time is below the 5 and 10 percentiles. Columns (4) re-estimates all the parameters including only trade flows from the following countries: Ecuador, Chile, Colombia, Panama, Costa Rica, Nicaragua, Guatemala, Mexico, Brazil, Argentina and Uruguay. Column (5) re-estimates all the parameters with average ocean transit times added to total and processing times only for trade flows from the following countries: China, United States, Germany, Italy, Spain and Brazil. Bootstrapped standard errors clustered by firm based on 500 repetitions are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

	New Importer	Experienced Importer
	(1)	(2)
	Estimation	
Processing Time	-0.269***	-0.164***
	(0.018)	(0.018)
Fixed Effect:		
Firm-Year		Yes
Origin-Product-Year		Yes
Observations		589,842
	Quantification	
χ	0.067***	0.041***
	(0.027)	(0.008)
arphi	2.020^{***}	2.058***
	(0.039)	(0.180)
ϑ	0.069^{***}	0.042***
	(0.017)	(0.008)
$r/(arphi-\omega)$	0.346***	0.120***
	(0.050)	(0.050)
$(\lambda - 1)$	0.117***	0.057***
	(0.033)	(0.013)
$(\lambda \cdot T^{\chi} - 1)$	0.283***	0.118***
	(0.011)	(0.010)

Table 10: Pro	cessing Costs	by Imp	orter Expe	rience
---------------	---------------	--------	------------	--------

The table reports IV estimates of variants of equation (8) that allows for different effects across types of firms: new importers (firms that never imported before) and incumbent importers (firms that have imported before). Firm-year and product-origin country-year fixed effects included (not reported). Standard errors clustered by firm are reported in parentheses below the estimated coefficients. In the case of the lower panel (Quantification), bootstrapped standard errors with 500 replications are reported. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

	Product Heterogeneity								
	Food and Beverage	Industrial Supplies	Capital Goods	Transport Equipments	Consumer Goods				
OLS $(\gamma \chi)$	-0.0916***	-0.0523***	-0.0439***	-0.0215	-0.0565***				
	(0.0297)	(0.00809)	(0.00816)	(0.0190)	(0.0105)				
IV $(\gamma \chi)$	-0.185^{***}	-0.256***	-0.231***	-0.187***	-0.265***				
	(0.0564)	(0.0217)	(0.0210)	(0.0454)	(0.0246)				
χ	0.0468***	0.0641***	0.0578***	0.0466^{***}	0.0662***				
	(0.000552)	(0.00438)	(0.00125)	(0.0165)	(0.000399)				
arphi	1.178^{***}	2.072^{***}	2.329^{***}	1.785^{***}	1.662^{***}				
	(0.00363)	(0.0195)	(0.0139)	(0.0341)	(0.0436)				
ϑ	0.0479^{***}	0.066^{***}	0.0584^{***}	0.0472^{***}	0.0687^{***}				
	(0.0007)	(0.0056)	(0.0014)	(0.032)	(0.0004)				
$r/(\alpha - \omega)$	0.274***	0 250***	0 237***	0 230***	0 392***				
$(\varphi \omega)$	(0.0028)	(0.0208)	(0.0155)	(0.0521)	(0.0369)				
$(\lambda - 1)$	0.099***	0 101***	0.088***	0.083***	0 131***				
(// 1)	(0.000)	(0.0104)	(0.000)	(0.0366)	(0.0013)				
$(\lambda \cdot T\chi = 1)$	0.185***	0 233***	0.196***	0 163***	0.3***				
(// 1 1)	(0.022)	(0.021)	(0.032)	(0.026)	(0.027)				

Table 11: Processing Costs by Product Categories

The table reports OLS and IV estimates of the same model as in Table 4. We estimate one specification for each product group. Firm-year and origin-product-year fixed effects included (not reported). Standard errors clustered by firm are reported in parentheses below the estimated coefficients. In the case of the lower panel (Quantification), bootstrapped standard errors with 500 replications are reported. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

11. Figures



Figure 1: Border Processing Time Distribution, 2013

Source: Authors' calculations based on data from SUNAT. The figure is a histogram of the processing time distribution. Data correspond to the year 2013.

A. Appendix - Tables

		Call	ao		
	Import	Number of	Number of	Number of	
Year	Value	Origins	Products	Employees	Age
2007	623.5	3.1	14.2	63.6	7.4
2008	785.1	3.0	13.2	60.4	7.4
2009	618.9	2.9	12.5	58.4	7.6
2010	660.8	2.9	12.7	58.1	7.7
2011	715.1	2.9	12.8	63.2	7.9
2012	700.5	2.9	12.8	64.8	8.0
2013	653.9	2.8	12.4	65.4	8.3
		All Im	ports		
37	Import	Number of	Number of	Number of	
Year	Value	Origins	Products	Employees	Age
2007	764.8	3.5	16.2	52.2	7.0
2008	1,009.3	3.3	14.8	48.4	7.0
2009	722.3	3.2	14.0	47.6	7.2
2010	904.8	3.2	14.2	47.8	7.3
2011	1,036.5	3.2	14.5	52.2	7.4
2012	1,057.4	3.2	14.4	52.2	7.5
2013	1,011.3	3.1	14.0	52.3	7.7
	Excluding	Minerals, Metals	and Air-Shipped	l Imports	
Vara	Import	Number of	Number of	Number of	A
rear	Value	Origins	Products	Employees	Age
2007	718.5	2.8	12.2	65.6	8.3
2008	657.1	3.1	14.1	63.6	7.4
2009	814.5	3.0	13.1	60.5	7.4
2010	629.2	2.9	12.5	57.8	7.6
2011	723.6	2.9	12.6	58.1	7.7
2012	796.3	2.8	12.6	63.2	7.9
2013	792.4	2.8	12.6	64.8	8.0

Table A1: Average Importer

Source: Authors' calculations based on data from SUNAT.

The table reports average import indicators for firms importing by sea through the Port of Callao, for all importers (including other ports), and for firms that do not import minerals, metals, or air-shipped goods. Import values are expressed in thousands of US dollars.

Stago	Channel	Avoraço	Percentile						
Stage	Ullalillei	Average	5th	10th	25th	50th	75th	90th	95th
Total Border Time	All	14.9	4	5	7	11	18	29	42
	Green	11.7	4	4	6	8	13	21	32
	Orange	16.1	6	7	9	13	20	28	39
	Red	23.4	8	9	13	19	28	42	56
Processing Time	All	4.3	1	2	2	3	6	12	16
	Green	1.8	1	1	2	2	3	5	6
	Orange	6	2	3	3	5	9	13	17
	Red	10.4	3	4	6	10	15	20	24
Storage Time	All	10.6	3	3	5	7	12	21	32
	Green	9.9	3	3	4	6	11	19	28
	Orange	10.1	3	3	5	7	12	20	28
	Red	13.1	3	3	5	8	15	28	41

Table A2: Border Times (in Days): Total and Stages in 2013, by Customs Verification Channel Excluding Express Shipments

The table reports the average and percentiles of the distribution of the total time to import, the total processing time, and storage time by customs verification channel (i.e., green, orange, and red) for 2013. The sample corresponds to all maritime imports entering into Peru through the port of Callao excluding shipments that enter under the express channel.

Table A3: Effect of Unloading Time on Storage Time - Excluding Express Channel Shipments

	(1)	(2)	(3)	(4)
Unloading Time	-0.174^{***}	-0.185***	-0.150***	-0.164***
	(0.017)	(0.019)	(0.016)	(0.018)
Firm Fixed Effect	Yes	No	Yes	No
Product-Origin Fixed Effect	Yes	No	Yes	No
Firm-Product-Origin Fixed Effect	No	Yes	No	Yes
Day Fixed Effects	No	No	Yes	Yes

Source: Authors' calculations based on data from SUNAT.

The table presents the effect of long unloading times on storage times conditional on fixed effects. The dependent variable is the natural log of storage time and the main explanatory variable is the natural log of the unloading time at the port. Standard errors clustered at importing firm-level are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level.

	Average	Standard Deviation
Total Imports	8.044	2.770
Trade Costs	0.107	0.136
Total Border Time	14.01	11.50
Processing Time	5.754	5.408
Channel	0.447	0.462
Congestion	5.424	1.985

Table A4: Summary Statistics for the Estimation Sample

Source: Authors' calculations based on data from SUNAT. The table reports average and standard deviation for the variables used in our regressions.

	(1)	(2)	(3)	(4)	(5)
Processing Time	-0.230***	-0.190***	-0.268***	-0.242***	-0.239***
	(0.013)	(0.013)	(0.017)	(0.034)	(0.020)
		First Stage			
Congestion	0.028***	0.027***	0.029***	0.028***	0.028***
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Channel	0.739^{***}	0.738^{***}	0.757^{***}	0.718^{***}	0.740^{***}
	(0.008)	(0.008)	(0.009)	(0.015)	(0.010)
F-Test	4,624.1	4,587.6	$3,\!910.9$	1,278.9	3,027.1
	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]
Fixed Effect:					
Firm-Year	Yes	No	No	No	Yes
Origin-Product-Year	No	Yes	Yes	Yes	Yes
Firm-Origin-Year	No	No	Yes	No	No
Firm-Product-Year	No	No	No	Yes	No
Firm-Product-Origin	No	No	No	No	Yes
Observations	589,842	$589,\!842$	$589,\!842$	$589,\!842$	$589,\!842$

Table A5: Robustness Checks: Alternative Fixed Effects

The table reports IV estimates of alternative specifications of equation (8) along with the first stage estimates and the effective F-test statistics. The dependent variable is the change in the natural log of the import value at the firm-product-origin-year level. The main explanatory variable is the change in the natural log of the median processing time. Standard errors clustered by importing firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

				IV			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Time	-0.184***	-0.180***	-0.167***	-0.187***	-0.156***	-0.174***	-0.153***
	(0.009)	(0.008)	(0.008)	(0.010)	(0.013)	(0.010)	(0.012)
		F	irst Stage				
Congestion	0.030***	0.030***	0.030***	0.030***	0.031***	0.030***	0.031***
	(0.0003)	(0.0002)	(0.0002)	(0.0003)	(0.0004)	(0.0003)	(0.0004)
Channel	0.682***	0.680***	0.681^{***}	0.685^{***}	0.682***	0.677^{***}	0.690***
	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)	(0.002)	(0.003)
Test Statistics							
F-Test	26726	31519	31207	23499	15917	19917	13088
	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]	[0.000]
Fixed Effects:							
Firm-Year	Yes	Yes	No	No	No	Yes	No
Origin-Product-Year	Yes	No	Yes	Yes	Yes	Yes	Yes
Carrier-Year	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Exporter-Year	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm-Origin-Year	No	No	No	Yes	No	No	No
Firm-Product-Year	No	No	No	No	Yes	No	No
Firm-Origin-Product	No	No	No	No	No	Yes	No
Firm-Origin-Product-Year	No	No	No	No	No	No	Yes
Exporting Firm-Year	No	No	No	No	No	No	Yes
Observations	685,971	685,971	685,971	685,971	685,971	685,971	685,971

Table A6: Robustness Checks: Exporting Firms and Carriers

The table reports IV estimates of alternative specifications of equation (8) along with the first stage estimates and the effective F-test statistics. The dependent variable is the change in the natural log of the import value at the firm-product-carrier-exporter-year level. The main explanatory variable is the change in the natural log of the median processing time. The instruments are port congestion as proxied by the median number of other vessels that arrive at the port the day before the vessel carrying the firm-product-carrier-exporter imports in question does in a given year and the average allocation to inspection. Columns correspond to different sets of fixed effects as indicated in the table. Standard errors clustered by importing firm are reported in parentheses below the estimated coefficients. * significant at the 10% level; ** significant at the 5% level; *** significant at the 1% level

A. Appendix - Theory

A.1. Proof Proposition 1

By observation, as long as $\varphi > \omega$, then $\frac{\partial t^*}{\partial \omega} > 0$. By the envelope theorem, $\frac{\partial ETC(t_l^*)}{\partial \omega} = \frac{t^{*^{\varphi}t_{\min}^{\varphi}rv}}{(\omega-\varphi)^2} > 0$. Also by the envelope theorem it is easy to observe that $\frac{ETC(t_l^*)}{\partial r} > 0$, because for any t_l^* the term $\frac{\varphi rv}{t_p} dt_p$ increases in r. By observation of equation (4), $\frac{\partial t_l^*}{\partial \omega} > 0$ and $\frac{\partial t_l^*}{\partial r} > 0$.

In order to prove that $\frac{\partial t^*}{\partial \theta} < 0$, we show that the semi-elasticity is negative. Taking logs on (4) and the partial derivative with respect to ϑ we obtain:

$$\begin{aligned} \frac{\partial \ln t}{\partial \vartheta} &= -\frac{\ln t_{\min}^{\varphi}}{[\vartheta + \varphi]^2} - \frac{1}{[\vartheta + \varphi]^2} \ln \left(\frac{r\varphi^2}{(\varphi - \omega)} \right) - \frac{[\vartheta + \varphi] \frac{1}{\vartheta} - \ln \vartheta}{[\vartheta + \varphi]^2} \\ \frac{\partial \ln t}{\partial \vartheta} &= -\frac{1}{[\vartheta + \varphi]^2} \left[\ln \left(\frac{t_{\min}^{\varphi} r\varphi^2}{(\varphi - \omega)\vartheta} \right) \right] - \frac{1}{\vartheta [\vartheta + \varphi]} \end{aligned}$$

Then $\frac{\partial \ln t^*}{\partial \theta} < 0$ as long as $\left(\frac{t_{\min}^{\varphi} r \varphi^2}{(\varphi - \omega) \theta}\right) > 1$. Imposing an interior solution then we can show that $\frac{r\varphi^2}{(\varphi-\omega)\vartheta} > 1$. Hence for $t_{min} \ge 1$ then $\frac{\partial \ln t}{\partial \vartheta} < 0$. The condition derived from the interior solution goes as follows:

$$t_{\min}^{\frac{\varphi}{\vartheta+\varphi}} \left(\frac{r\varphi^{2}}{(\varphi-\omega)\vartheta}\right)^{\frac{1}{\vartheta+\varphi}} > t_{\min}$$

$$t_{\min}^{\frac{\varphi}{\vartheta+\varphi}-1} \left(\frac{r\varphi^{2}}{(\varphi-\omega)\vartheta}\right)^{\frac{1}{\vartheta+\varphi}} > 1$$

$$t_{\min}^{\frac{-\vartheta}{\vartheta+\varphi}} \left(\frac{r\varphi^{2}}{(\varphi-\omega)\vartheta}\right) > 1$$

$$t_{\min}^{\frac{-\vartheta}{\vartheta+\varphi}} \left(\frac{r\varphi^{2}}{(\varphi-\omega)\vartheta}\right) > 1$$

$$\left(\frac{r\varphi^{2}}{(\varphi-\omega)\vartheta}\right) > t_{\min}^{\frac{\vartheta}{\vartheta+\varphi}} > t_{\max}^{\frac{\vartheta}{\vartheta+\varphi}} > t_{\max$$

1

Appendix - Figures

Figure A.1: Import Product Shares, Callao versus U.S. (Source Data: UN Comtrade Database, Copyright ©United Nations 2012, and, SUNAT)

